

Temi di discussione

(Working papers)

Values, inequality and happiness

by Claudia Biancotti and Giovanni D'Alessio



The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Domenico J. Marchetti, Patrizio Pagano, Ugo Albertazzi, Michele Caivano, Stefano Iezzi, Paolo Pinotti, Alessandro Secchi, Enrico Sette, Marco Taboga, Pietro Tommasino.

Editorial Assistants: ROBERTO MARANO, NICOLETTA OLIVANTI.

VALUES, INEQUALITY AND HAPPINESS

by Claudia Biancotti* and Giovanni D'Alessio*

Abstract

This paper examines the relationship between inequality and happiness through the lens of heterogeneous values, beliefs and inclinations. Drawing upon opinion data from the European Social Survey for twenty-three countries, we find that individual views on a wide range of themes can be effectively summarized by two orthogonal dimensions: moderation and inclusiveness. The former is defined as a tendency to take mild stands on issues rather than extreme ones; the latter is defined as the degree of support for a social model that grants equal rights to everyone who willingly subscribes to a shared set of rules, regardless of background and circumstances. These traits matter when it comes to how inequality affects subjective well-being; specifically, those who are either more moderate or more inclusive than their average compatriots prefer lower levels of inequality. In the case of moderation, inequality aversion can be read in terms of a desire for stability: people who are reluctant to take strong stands are especially wary of conflict, tension and unrest, which often go hand-in-hand with disparities. In the case of inclusiveness, the main element at play is likely to be distress accruing on a perception of unfairness.

JEL Classification: D31, D63.

Keywords: happiness, inequality, heterogeneity.

Contents

| 1 | Introduction | 3 |
|-----|----------------------|----|
| 2 | The literature | 4 |
| 3 | The model | 7 |
| 4 | The data | 9 |
| | 4.1 Income variables | 11 |
| | 4.2 Values | 13 |
| 5 | Results | 16 |
| 6 | Conclusions | 21 |
| Ref | ferences | 22 |
| Ap | pendix | 27 |

^{*} Bank of Italy, Economic and Financial Statistics Department.

1 Introduction

Is inequality desirable or undesirable? Does it have an effect on well-being, and is this effect uniform across countries and people? These are very important research questions, if only on account of the political relevance of debates on redistribution.

Different disciplines provide different answers. Economists generally seek to qualify inequality according to its effect on indicators of material welfare, such as the growth rate of GDP per capita; so far, results have been mixed [Alesina and Perotti, 1996; Bertola, 1999; Forbes, 2000; Keefer and Knack, 2002; Quah, 2003; Banerjee and Duflo, 2003]. Moral philosophers and social choice scholars, on the other hand, tend to evaluate income distributions based on their compliance with a theory of justice, again with varied conclusions [Rawls, 1971; Dworkin, 1981; Cohen, 1989; Lefranc, Pistolesi and Trannoy, 2006]. The two camps overlap only occasionally; when they do, discussions arise over the relative importance that should be attributed to economic performance and ethical concerns [World Bank, 2005; Roemer, 2006].

In recent years, a new perspective has emerged. Data on subjective well-being or 'happiness', once used almost exclusively in psychology, have gained credibility among economists in the wake of the success enjoyed by behavioural studies, and of improved techniques for the measurement of subjective states [Frey and Stutzer, 2002; Kahneman and Kruger, 2006; Di Tella and McCulloch, 2006]. Researchers can now ascertain how people feel about inequality and why, as a useful complement to asking whether a distributive rule is good or bad in the light of a general principle.

The happiness perspective appears to offer a major advantage in the analysis of a political and emotional wedge issue such as inequality: it is intrinsically focused on heterogeneity. Different individuals have been found to map the same events of life into happiness levels in a different way, depending on circumstances, attitudes, beliefs, and a host of other factors [Rojas, 2007]¹. In a sense, happines studies are naturally positioned at the

¹For example, Frey and Stutzer [2005] show that a worker's ethnic background is important in determining how workplace policies in favour of racial integration affect their happiness. Becchetti, Castriota and Giuntella [2006] use data on subjective well-being to estimate how the trade-off between inflation and unemployment differs across social groups, following the idea of community group indifference maps originally provided by Chossudovsky [1972]. McFarlin and Rice [1992] prove that subjective facet importance is

final stage of the debate, already incorporating the idea that the ordering of distributions is better left to politics, at least until the weight assigned to notions such as equity or efficiency varies across the population.

This paper brings together in a simple model suggestions from different strands of thought. We hypothesize that inequality exerts an effect on subjective well-being, and focus on the role of personal values as a filter between the two. Looking at survey data for more than 32,000 individuals in twenty-three European countries, we perform multivariate analysis on eighteen statements related to diverse moral issues, and extrapolate two orthogonal dimensions: moderation, defined as a tendency to take mild stands on issues rather than extreme ones, and inclusiveness, defined as support for a social model that grants equal rights to everyone who willingly subscribes to a shared set of rules, regardless of background and circumstances. When regressing happiness on the interaction between values and distributive indicators, we find that individuals who are either more moderate or more inclusive than their representative compatriots prefer lower levels of inequality. In the case of moderation, we read this result in terms of a desire for stability: people who are reluctant to take strong stands are especially wary of conflict, tension and unrest, frequent companions to disparities. In the case of inclusiveness, the main element at play is likely to be distress accruing on a perception of unfairness.

The paper is structured as follows. Section 2 provides a review of the literature on inequality and happiness. Section 3 introduces our model. Section 4 describes the data. Section 5 presents the results. Section 6 concludes. The Appendix contains further tables.

2 The literature

Many papers have been written on the relationship between inequality and happiness, usually incorporating the idea that the connection is strongly subjective in nature. The individual filter has been analysed from a number of angles.

The approach closest to the one we propose here singles out the role played by interpretation: happiness is not influenced by the distribution of

a non-negligible factor in determining overall levels of job satisfaction. Kohler, Behrman and Skytthe [2005] find that the impact of parenthood on happiness can be different for mothers and fathers.

income *per se*, but rather by the social ordering observers read into it. Most authors highlight the relevance of cultural elements in shaping the interpretation process, thus falling into step with the growing literature devoted to how collective beliefs affect economic behaviours and outcomes [Guiso, Sapienza and Zingales, 2003; Tabellini, 2006; Férnandez and Fogli, 2006; Horii, Jin and Levitt, 2005].

Graham and Felton [2006] find that in Latin America the correlation between Gini coefficients and happiness is strongly negative, since most people perceive it as an indicator of persistent social rifts and poverty traps. Alesina, Di Tella and McCulloch [2004] analyse the differences between North America and Europe, also taking into account heterogeneity in social class and political positioning. They conclude that the American poor do not dislike inequality, because they believe that adequate effort will save them; the American rich who define themselves as left-leaning do, because they find it unfair, and possibly because they are afraid of losing their place. In Europe, on the contrary, aversion to inequality is found chiefly among the poor, who feel stuck in their situation independent of political orientation. While not directly tackling subjective well-being, Bénabou and Tirole [2006] present similar insights in their work on beliefs and redistributive policies: North Americans mostly see poverty as an indicator of laziness, and inequality as a signal of mobility, while Europeans tend to believe that the poor are mainly unlucky, and associate inequality with unearned fortunes.

Another popular point of view in the study of inequality and happiness is the positional one: people do not like to see inequality because it makes them feel bad about their own circumstances relative to others. The relatively poor resent their economic inferiority (envy), and the relatively rich resent their enjoyment of privileges that others do not have access to (guilt). In their seminal paper on cooperative behaviour in games, Fehr and Schmidt [1999] term the combination of these feelings 'self-centred inequity aversion'.

The theme of envy has illustrious precedents: the idea that upward comparison leads to dissatisfaction dates as far back as Veblen [1899], and was greatly expanded upon by Easterlin [1974] and Hirsch [1976] in their studies on adaptive expectations and status signalling. If we assume that individual desire for consumption is influenced by social standards, then the comparison with others becomes, to a certain extent, also a comparison with one's own aspirations; psychology offers a class of 'have-want' models for the analysis of this phenomenon, recently adopted by economists interested in happiness [Stutzer, 2004; Easterlin, 2006]. Clark and Oswald [1996] find that the self-reported satisfaction of British workers is negatively related to their benchmark wage, i.e. the average wage earned by workers with the same qualifications and experience; Luttmer [2005] has similar results for the United States.² As for the role of guilt as an engine of inequality aversion on the part of the rich, results in behavioral economics appear to prove that downward comparison might curb happiness because of various factors, including but not limited to feelings of altruism, a yearning for justice, fear that a privileged place in society can make one the target of resentment and violence, and even an interest in allocative efficiency [Hoffman, McCabe and Smith, 1996; Fehr and Fischbacher, 2002; Henrich *et al*, 2004].

In this paper, we take into account both interpretation processes and positional concerns. Although the two have already been considered together in experiments based on modified Ultimatum Games [Hoffman, McCabe, Shachat, and Smith, 1994], to our knowledge there is only one work that does so in the context of the relationship between inequality and happiness: Clark [2003] estimates the effects of both relative income and absolute inequality in Britain, finding that happiness is increasing in both (people probably approve of competing, but disapprove of losing).

Moreover, while the importance of beliefs has been singled out in the past, our contribution appears to be the first that articulates the relationship between a wide array of core moral values and feelings on inequality at the individual level. The micro-level perspective, while common in sociology [Inglehart and Baker, 2000], so far has not been used frequently by economists, who prefer to typify individuals based on exogenous partitions that function as proxies for inherited culture and customs: religious denomination, nationality, ethnicity. There are valid reasons for this choice: most importantly, the set of priors and preferences accruing to a certain type of religious doctrine or to a certain national community has been shown to be so slow-moving that reverse causation from almost any dependent variable to those features can be ruled out, at least in the short run. We try simultaneously to preserve the high degree of freedom and detail allowed by

²There are even indications that this type of feeling might not be exclusive to humans, as the experiments on monkeys conducted by Brosnan and de Waal [2003] show: capuchin monkeys trained to exchange rock tokens for food react badly whenever they perceive that they are being shortchanged compared with their peers.

micro-level analysis and to avoid the problems of causality by limiting our study to beliefs and preferences that appear to be set deep in the psychological makeup of individuals or in their upbringing, and should therefore not be influenced by the level of happiness experienced in a given period.

3 The model

The model we propose integrates the suggestions of the literature discussed above with a systematic attempt to formalise the precise nature of the process through which agents filter perceptions of inequality into feelings of happiness.

The idea is as follows. Individuals look at the whole distribution of incomes in order to determine their well-being; in particular, they take into consideration, besides their own income, one or more measures of position, and one or more measures of dispersion. The former constitute a reference point for the determination of one's social status, while the latter are read as a description of the social ordering and its degree of persistence. Distributive features are then filtered into happiness on the basis of inclinations, beliefs and values, defined respectively as innate character traits, priors about how the world actually works, and preferences about how it should work [Guiso, Sapienza, and Zingales, 2006]. For the sake of simplicity, we will refer to these three concepts with the sole term 'values' in the following. In particular, we want to see whether heterogeneity in personal values implies heterogeneity in the links between inequality and happiness.

The idea can be formalized as follows:

$$H_i = h(g(f_i(x), v_i), q_i) \tag{1}$$

where H_i is the happiness level for agent *i* and *h* is the technology of happiness production. The function *g* describes how v_i , the vector of personal values for individual *i*, interacts with the perceived density function of income $f_i(x)$ in order to produce a judgement on distributive matters. Finally, q_i is a vector of known determinants of happiness, such as health and marital status.

From (1) we derive the following equation for estimation purposes:

$$\mathbf{H} = \beta [\mathbf{S} - \overline{\mathbf{S}}_c]' [\mathbf{Y} \ \overline{\mathbf{Y}}_r \ \overline{\mathbf{Y}}_c \ \mathbf{G}_r] + \gamma \mathbf{Q} + \varepsilon$$
(2)

where **H** is the vector of self-reported happiness levels; $[\mathbf{S} - \overline{\mathbf{S}}_c]$ is a matrix of synthetical indicators of personal values, resulting from multivariate analysis of elementary value items and expressed as deviations from national medians; $\mathbf{Y}, \overline{\mathbf{Y}}_r, \overline{\mathbf{Y}}_c$ and \mathbf{G}_r are respectively the vectors of personal incomes, median incomes in the respondent's region and country, and standardized interquartile ranges for incomes in the respondent's region. Finally, \mathbf{Q} is a matrix of controls.

We choose to estimate a few synthetical indicators from a large array of elementary value items, rather than directly picking a group of values small enough to generate an economical specification of the regression. We want to eschew self-fulfilling intuitive priors, such as the idea that a favourable disposition to solidarity matters more than, say, the level of trust towards others in determining what kind of income distribution a person desires.

Values are expressed in terms of deviations from the national median because we know, from results in public choice theory first derived by Meltzer and Richard [1981] and subsequently refined, that the distribution of income is influenced by the institutional and political features of a country. Those, in turn, can be held to reflect prevailing preferences through electoral mechanisms, a particularly relevant aspect for a sample consisting entirely of democratic countries. While the derivation of a voting model is beyond the goal of this paper, we need to neutralize potential endogeneity problems arising from the correlation between mainstream opinions and observed inequality levels; looking at relative rather than absolute values appears to be the most natural solution. In doing so, we also follow the insight offered by psychologists Sagiv and Schwartz [2000], according to whom 'well-being [also] depends upon congruence between personal values and the prevailing value environment'. Our reference unit for calculating deviations is the country because, to the best of our knowledge, the large majority of welfare policies in the area we consider are decided at the national level.

Where income is concerned, we use positional and distributive indicators at the regional level instead, so as to better represent the idea of *perceived* income distribution expressed in (1). We follow an insight that is present throughout the literature reviewed in Section 2: people are imperfectly informed about the income of others, and in general the knowledge that person A has about the standard of living of person B is inversely proportional to the geographical distance between the two, both because of direct exposure and the impact of local news media. This idea is incorporated in our model by assuming that people derive their distributive facts from short-range information only.³ We add the national median of incomes as a proxy of the quality of public goods and the general standards of living.

4 The data

The empirical analysis is based on data from the second round of the European Social Survey, carried out in 2004. The Survey, funded by the European Commission, the European Science Foundation and several national partners, 'has been mapping long-term attitudinal and behavioural changes in Europe's social, political and moral climate' since 2001 [www.europeansocialsurvey.org, 2007]. It is directed by an international Central Co-ordinating Team based in London, and carried out every two years by independent national teams; a single questionnaire is created in English, then translated into several languages. Contrary to other surveys of a similar nature, the sampling design is entirely probabilistic. In countries where lists of households are available, the sampling unit is the household; otherwise, the sampling unit is the street address, then a household living at that address is chosen at random. In both cases, the final respondent is an adult chosen at random among the members of the household.

Each round has a core module covering twelve broad topics, ranging from demographics and financial circumstances to political engagement and subjective well-being. Several rotating modules, which vary from round to round, complement the core module. We focus on the second round only, disregarding the first, not only because it covers a larger number of countries,

³This strategy ignores the suggestion provided by public choice theory according to which perceptions such as the one we are studying should be reconstructed based both on where an agent lives and on their degree of interest in the phenomenon at hand: someone who follows economic news closely might have a precise idea of the national and even international distribution of income, while someone who is uninterested in the matter might have a knowledge that is limited to the immediate neighbourhood. The hypothesis has been proven correct in several occasions, but we cannot employ it in our empirical work for two reasons. Even if we were able to identify groups with different informational scopes, which may be possible in the light of available data, the sample size would not allow us to estimate inequality at the appropriate level for the less informed, say town or district. Also, we would need assumptions concerning not only the relationship between the self-reported degree of information and the scope of knowledge about incomes, but also the distribution of the error term, which would probably be both higher and more variable for the less informed. These two processes appear to introduce a degree of arbitrariness that offsets the gains.

but also because it offers a rotating module on economic morality, which helps to ascertain individual attitudes towards a wide range of economic behaviours. At the time of writing this paper, data for twenty-six countries had been released to the public; we included twenty-three.⁴ Most national samples comprise between 1,500 and 2,500 observations, for a total of 43,650, and all come with design weights for national estimates. For Europe-wide estimates, population weights are provided that correct for the imbalance in sampling fractions.

Item non-response is a serious problem in the ESS; data on income and on personal values, both of which are essential for our model, are particularly affected. We tried to balance quality and quantity through a mixture of model-based imputation, variable selection and data deletion, as discussed in Sections 4.1 and 4.2. The final sample includes 35,335 observations, or 80.95 per cent of the original ESS sample (Tables 1 and A.1), with national samples ranging from 507 households for Iceland to 2,370 households for Germany.

| Country | Ν | $\begin{array}{c} \mathbf{Sampling} \\ \mathbf{fraction} \\ (^0\!/_{\!00}) \end{array}$ | Country | Ν | $\begin{array}{c} \mathbf{Sampling} \\ \mathbf{fraction} \\ (\%_0) \end{array}$ |
|----------------|-----------|---|----------------|------------|---|
| Austria | $1,\!602$ | 0.197 | Luxembourg | 1,295 | 2.868 |
| Belgium | $1,\!636$ | 0.157 | Netherlands | 1,745 | 0.107 |
| Czech Republic | 1,781 | 0.174 | Norway | 1,710 | 0.374 |
| Denmark | 1,248 | 0.231 | Poland | 1,307 | 0.034 |
| Estonia | 1,289 | 0.954 | Portugal | 1,626 | 0.155 |
| Finland | 1,887 | 0.362 | Slovakia | 1,022 | 0.190 |
| France | $1,\!404$ | 0.023 | Slovenia | 1,152 | 0.577 |
| Germany | $2,\!430$ | 0.029 | Spain | 1,321 | 0.031 |
| Greece | 2,048 | 0.185 | Sweden | 1,754 | 0.195 |
| Hungary | $1,\!153$ | 0.114 | Switzerland | 1,860 | 0.253 |
| Iceland | 507 | 1.745 | United Kingdom | 1,755 | 0.029 |
| Ireland | 1,803 | 0.448 | Total | $35,\!335$ | 0.087 |

Table 1: Sample size and sampling fraction, by country⁺

⁺Sampling fractions are computed based on population statistics for 1.1.2004, as provided by Eurostat in *Population and Social Conditions*, 2005/15.

The ESS questionnaire can be found on the Web, along with methodological documentation. In the following, we skip a detailed discussion of variables that were merely taken in their original state and inserted in our estimates as controls; instead, we focus on the treatment of income variables,

⁴Italy, Turkey and Ukraine were excluded on account of heavy item non-response for questions related to values and beliefs.

and on the choice of indicators for beliefs and values.

4.1 Income variables

The ESS questionnaire features the following item:

[I]f you add up the income from all sources, which letter describes your household's total net income? If you don't know the exact figure, please give an estimate.

Respondents are shown a card listing twelve brackets of weekly income, each labelled with a different letter; the scale is also converted to monthly and yearly equivalents for the sake of clarity. The brackets are of unequal size, smaller at the bottom and larger at the top; the extreme ones are openended, respectively including any income below 40 euros per week and any income above 2,310 euros per week.

The item non-response rate for this question totals 19.61 per cent of the final sample and is unevenly distributed across countries: in Norway and Sweden it is below 3 per cent, while in Portugal and Greece it exceeds 30 per cent. Since the willingness to provide income information is very likely to be correlated with culture and values, as suggested on an intuitive level by the geographical distribution of response rates, the mere elimination of observations with missing values would probably introduce selection bias and distort the results of subsequent analyses.

For each country, we estimate a simple logistic regression linking income class with household size and with the answer to the following question:

Which of the [se] descriptions comes closest to how you feel about your household's income nowadays: living comfortably on present income, coping on present income, finding it difficult on present income, or finding it very difficult on present income?

The model turns out to have good explanatory power for all countries, with the share of concordant observation-prediction pairs ranging from 68.2 to 87.0 per cent depending on the country. We therefore employ it to impute missing values: the resulting distribution is close to the original one (Table 2).

Even after taking care of missing data we still are left with income classes, not income levels: the only information available on those is limited to the

| Class | Abso | olute freque | te frequencies Relative frequencies | | | |
|-----------------|---------|----------------|-------------------------------------|---------|----------------|-----------------|
| Class | Imputed | Not Imputed | Total sample | Imputed | Not Imputed | Total sample |
| 1 | 96 | 409 | 505 | 1.39 | 1.44 | 1.43 |
| 2 | 331 | 1,394 | 1,725 | 4.78 | 4.91 | 4.88 |
| 3 | 608 | 2,176 | 2,784 | 8.77 | 7.66 | 7.88 |
| 4 | 1,185 | 4,068 | 5,253 | 17.10 | 14.32 | 14.87 |
| 5 | 919 | 4,033 | 4,952 | 13.26 | 14.20 | 14.01 |
| 6 | 804 | 3,255 | 4,059 | 11.60 | 11.46 | 11.49 |
| 7 | 645 | 2,772 | 3,417 | 9.31 | 9.76 | 9.67 |
| 8 | 678 | 2,799 | 3,477 | 9.78 | 9.85 | 9.84 |
| 9 | 980 | 4,540 | 5,520 | 14.14 | 15.98 | 15.62 |
| 10 | 406 | 1,941 | 2,347 | 5.86 | 6.83 | 6.64 |
| 11 | 147 | 605 | 752 | 2.12 | 2.13 | 2.13 |
| 12 | 132 | 412 | 544 | 1.90 | 1.45 | 1.54 |
| Sample total | 6,931 | 28,404 | 35,335 | 100.00 | 100.00 | 100.00 |
| Row frequencies | - | - | - | 19.62 | 80.38 | 100.00 |

Table 2: Count and distribution of income classes

five per cent of the original sample who answered a question on the individual net pay of the respondent alone for his or her main occupation. However, classes are not adequate for the estimation of position and dispersion measures, and neither are simple imputation procedures based on random draws from uniform distributions within the classes. First, the distribution of income is generally log-normal, implying that the distribution within classes is skewed to the left for low incomes and to the right for high incomes; the assumption of uniformity results in (probably asymmetrical) overestimation of the weight on distribution tails, yielding in turn unpredictable effects on inequality measures. Also, we need to take demographic structure into account, especially because our sample includes both Western and Eastern European countries; if, say, larger households are routinely closer to the upper bound of their income class than smaller households, neglecting household size and composition will lead to underestimation of incomes in countries with higher fertility rates.

In order to take these important issues into account, we undertake a further imputation step based on stratified density estimation. For each country and for each household size a Gaussian kernel of the whole distribution is estimated; the resulting density is then normalized within each class, so that household incomes can be drawn at random. In order to perform this operation in the extreme classes, we apply bottom-coding and top-coding: the lowest bracket is closed at null income and the upper bracket is closed at 150,000 euros.⁵Amounts thus obtained are equivalized with the modified OECD equivalence scale. The weighted mean, median and standardized interquartile range for the distribution of equivalent incomes are then computed both at the national level and separately for each region, following the EU-NUTS2 partition where the data allows, and the EU-NUTS1 partition otherwise.

4.2 Values

The ESS offers hundreds of value-related questions, but most of them are only answered by a share of the sample. We need to select a subset of variables that strike a reasonable balance between the richness of individual information and the availability of valid observations. Once this goal is attained, a small number of synthetic indicators must be produced so as to provide convenient input for regression analysis.

We go about the first task by classifying available information on beliefs and values into six broad thematic categories: trust, solidarity, legality, civic engagement, the family, and diversity. Within each domain, the three variables with the lowest incidence of item non-response are chosen, for a total of eighteen variables (Table A.2). The eighteen items are then subjected to multiple correspondence analysis, a form of multivariate analysis suitable for qualitative data. This technique is essentially a version of principal component analysis based on the chi-square metric rather than on the euclidean metric; for details see Benzécri [1973] and Lebart, Morineau and Warwick [1984]. In layman's terms, we study how opinions expressed by individuals on a large set of topics combine along a limited number of orthogonal dimensions called factors, which are entirely endogenous to the data, and should serve as a way of summarizing and interpreting them.

The estimation of multiple correspondences yields a reasonably good fit, with the first two factors explaining 79.4 per cent of the total variance generated by the eighteen individual variables (Table 3). Factor loadings are reported in Table A.3, complete with relevant fit statistics.

Factor 1 explains 48.9 per cent of total variance. For the question 'Gen-

⁵Different choices, including the use of country-specific brackets and alternative kernel functions, have been subjected to testing and do not appear to exert any considerable influence on the final estimates.

| Factor | Inertia | Adjusted inertia | Variance explained | Cumulative variance | Goodness of fit |
|--------|---------|---------------------|-----------------------|------------------------|-----------------|
| | | (Benzécri) | (per cent) | explained | |
| 1 | 0.16374 | 0.01312 | 48.88 | 48.88 | ***** |
| 2 | 0.14101 | 0.00819 | 30.5 | 79.38 | ***** |
| 3 | 0.09443 | 0.00169 | 6.31 | 85.69 | *** |
| 4 | 0.08447 | 0.00094 | 3.49 | 89.18 | ** |
| 5 | 0.08029 | 0.00069 | 2.56 | 91.74 | * |
| 6 | 0.07641 | 0.00049 | 1.82 | 93.55 | * |
| 7 | 0.07391 | 0.00038 | 1.41 | 94.96 | * |
| 8 | 0.07245 | 0.00032 | 1.19 | 96.15 | * |
| 9 | 0.07198 | 0.00030 | 1.13 | 97.28 | * |

Table 3: Inertia and explained variance for multiple correspondence analysis $^+$

+Only factors that explain one per cent or more of global variance are included in the table.

erally speaking, would you say that most people can be trusted, or that you can't be too careful?', for which an 11-point response scale is offered where 0 means 'You can't be too careful' and 10 means 'Most people can be trusted', positive factor loadings are estimated on scores of 4 to 9; they are highest for scores of 6 and 7. Negative loadings appear for very low scores and for the top score (Figure 1). For the question 'When jobs are scarce, men should have more right to a job than women', positive loadings are observed for 'Agree', 'Neither agree nor disagree', and 'Disagree', with the latter option also being the one with the highest estimate; negative loadings are found for 'Strongly agree' and 'Strongly disagree'. In the case of 'How wrong is it for someone to sell something second-hand and conceal some or all of its faults?', positive loadings again apply for the intermediate response options 'A bit wrong' and 'Wrong', while 'Not wrong at all' and 'Seriously wrong' are associated with negative estimates. A similar U-shaped profile for loadings emerges for the five-point agreement scale proposed for 'Society would be better off if everyone just looked after themselves', for the four possible answers to 'To what extent do you think your country should allow people of a different race to come and live here?', and for nearly all other questions.

The picture painted by the analysis of loadings suggests that Factor 1 can be tentatively interpreted as an indicator of *moderation*, defined as the tendency to express mild opinions rather than extreme ones: individuals who score high on this factor are more likely to report agreement, disagreement or (less frequently) lack of opinion with respect to any given statement



Figure 1: An example of factor loadings

than to express strong agreement or strong disagreement. The inclination towards moderation measured by the factor turns out to be independent of the specific beliefs held: negative loadings are consistently estimated for extreme values of the agreement scale on items as different as 'A woman should be prepared to cut down on her paid work for the sake of her family' and 'Gay men and lesbians should be free to live as they wish'.

Factor 2 explains 30.5 per cent of total variance. Positive factor loadings are associated with high levels of trust (score from 6 or above), disagreement or strong disagreement with a men-first policy in the job market, thorough condemnation of fraud, disagreement or strong disagreement with the idea that society would be better if everyone looked after themselves, and openness towards immigrants of different races. Individuals who score high on this axis also believe that citizens should spend some of their free time helping others, state that they are not afraid of being treated dishonestly, appear supportive of gender equality from a number of perspectives, and actively participate in politics.

The joint consideration of these elements hints at a possible interpretation of Factor 2 in terms of *inclusiveness*, defined as support for the extension of rights and opportunities to everyone, regardless of background and circumstances. There is also an element of social cohesion based on the consistent subscription to a shared set of rules.





Figure 2 represents country medians on the two factor axes. Scandinavia scores high on both moderation and inclusiveness, while Mediterranean countries and Eastern European ones are located in the opposite quadrant. Central European and Anglophone countries display positive scores for moderation, and hover around the mean with respect to inclusiveness.

The analysis of the correlation matrix between factor scores and other individual traits reveals that associations between values, demographics and social class are weak, with the Pearson correlation coefficient reaching a maximum of 0.39 in the case of inclusiveness and the number of years spent in formal education. Age seems to have no correlation with moderation (-0.07) and a slightly negative correlation (-0.14) with inclusiveness. Income also feebly correlates with both (0.12 and 0.23 respectively).⁶

5 Results

We want to test whether heterogeneity in values, as summarized in our two-factor representation, also implies heterogeneity in the links between

⁶In the case of income, correlations are bound to be artificially weakened by the imputation process. When computed on just raw data, however, the coefficients change only slightly.

inequality and happiness. To this end, we run a set of ordered logits where the degree of happiness in a scale ranging from 0 to 10 appears on the lefthand side, while a number of variables related to the distribution of income at the regional level appear on the right-hand side, both in raw form and interacted with deviations from national medians in terms of moderation and inclusiveness. These core variables are supplemented by a large number of controls that are routinely used in happiness regressions.

Table 4 presents our results. When estimating the specification in (2), the regression coefficient for our indicator of inequality amounts to 0.24, and after correction of standard errors for clustering at the regional level it is barely significant at the 10 per cent level. However, the interaction between the same indicator and individual deviation in moderation from the national median has a coefficient of -1.27, significant at the 1 per cent level; the interaction between inequality and individual deviation in inclusiveness from the national median has a coefficient of -0.63 and is significant at the 5 per cent level.⁷

| | Estimate | Std Err⊤ | Pr>ChiSq |
|--|--------------|--------------|----------|
| Equivalent income | | | |
| Log own | 0.144 | 0.024 | 0.000 |
| *Moderation (deviation from median) | -0.081 | 0.070 | 0.248 |
| *Inclusiveness (deviation from median) | -0.160 | 0.073 | 0.028 |
| Std interquartile range, regional | 0.248 | 0.150 | 0.098 |
| *Moderation | -1.272 | 0.282 | 0.000 |
| *Inclusiveness | -0.627 | 0.307 | 0.041 |
| Log median, national | 0.466 | 0.171 | 0.006 |
| *Moderation | -0.026 | 0.252 | 0.917 |
| *Inclusiveness | -0.469 | 0.330 | 0.156 |
| Log median, regional | -0.105 | 0.165 | 0.525 |
| *Moderation | -0.201 | 0.253 | 0.426 |
| *Inclusiveness | 0.225 | 0.349 | 0.520 |
| Demographics | 1 | 1 | ł |
| Gender: female | 0.115 | 0.038 | 0.002 |
| Age | -0.066 | 0.009 | 0.000 |
| Age squared | 0.001 | 0.000 | 0.000 |
| Self-reported health (baseline = very | good) | • | |
| Good | -0.578 | 0.053 | 0.000 |
| Fair | -1.062 | 0.075 | 0.000 |
| Bad | -1.756 | 0.114 | 0.000 |
| Very bad | -2.447 | 0.334 | 0.000 |
| Marital status (baseline = married or | in registere | d cohabitati | ion) |

Table 4: Ordered logit estimates for happiness

⁷Table A.4 in the Appendix shows that when values are not considered the effect of inequality on happiness is not significant.

| | Estimate | Std Err ⁺ | Pr>ChiSq |
|---------------------------------------|-------------|----------------------|----------|
| Separated | -0.964 | 0.148 | 0.000 |
| Divorced | -0.635 | 0.073 | 0.000 |
| Widowed | -0.927 | 0.076 | 0.000 |
| Never married | -0.647 | 0.049 | 0.000 |
| Children | • | • | · |
| Children living at home | -0.058 | 0.049 | 0.233 |
| Children living outside the home | 0.146 | 0.051 | 0.004 |
| Social ties | | | |
| At least one close friend | 0.588 | 0.080 | 0.000 |
| Frequency of social activity | 0.157 | 0.017 | 0.000 |
| Location (baseline $=$ city center) | | • | |
| Suburbs or outskirts of big city | -0.004 | 0.072 | 0.957 |
| Town or small city | 0.065 | 0.062 | 0.291 |
| Country village | 0.125 | 0.063 | 0.047 |
| Farm or home in countryside | 0.288 | 0.108 | 0.008 |
| Feeling of safety in own neighbourhoo | d (baseline | = very safe) | |
| Safe | -0.084 | 0.043 | 0.053 |
| Unsafe | -0.180 | 0.061 | 0.003 |
| Very unsafe | -0.332 | 0.115 | 0.004 |
| Job status (baseline $=$ employee) | • | • | |
| Student | 0.060 | 0.080 | 0.453 |
| Unemployed, looking for job | -0.558 | 0.116 | 0.000 |
| Unemployed, not looking | -0.374 | 0.148 | 0.011 |
| Permanently sick or disabled | 0.085 | 0.139 | 0.540 |
| Retired | 0.168 | 0.067 | 0.012 |
| Community or military service | 0.258 | 0.279 | 0.355 |
| Housework | 0.051 | 0.066 | 0.439 |
| Other | -0.206 | 0.183 | 0.259 |
| Other factors | | | |
| Years in formal education | -0.002 | 0.006 | 0.775 |
| Intensity of religious belief | 0.054 | 0.009 | 0.000 |
| Belongs to discriminated group | -0.538 | 0.091 | 0.000 |
| Homeowner | 0.102 | 0.044 | 0.021 |
| Marginal effect of deviation from the | median in v | alues | |
| Moderation | 3.922 | 0.715 | 0.000 |
| Inclusiveness | 5.062 | 1.065 | 0.000 |
| Model fit statistics: | | | |
| Prob > Chi Square (Wald) | | 0.000 | |
| Pseudo \mathbb{R}^2 | | 0.061 | |

Table 4: Ordered logit estimates for happiness (continued)

+Standard errors are adjusted for clustering at the regional level.

In other words, people who are more moderate or more inclusive than their fellow citizens tend to dislike inequality more. Since we are discussing moral traits, it is hard to put a strict quantitative interpretation on the absolute values of regression coefficients; but the deviations from national medians in terms of moderation and inclusiveness have similar standard deviations, respectively estimated at 0.38 and 0.35, and it is therefore reasonable to say that, for a given relative distance from the mainstream, increases of inequality have an effect more or less twice as bad on those who are distant because of moderation than on those who are distant because of inclusiveness.

The result can be interpreted as follows: those who are more moderate than their compatriots might dislike inequality because it acts as a trigger for social tension, conflict and unrest, as described by the literature referenced in Section 1; those who are particularly inclusive might dislike inequality because they perceive it as morally unfair. The former motive, known in the literature as instrumental inequality aversion, appears to be stronger than the latter, i.e. substantive inequality aversion.

Values also appear to be significant when it comes to deriving happiness from one's own income. The net effect of the logarithm of equivalent income is positive, but turns negative when interacted with inclusiveness: this may be an example of the guilt effect outlined in Section 2. The interaction elements are not significant when it comes to the impact exerted on happiness by the general standard of living, as expressed by the national median of equivalent incomes: the marginal effect is positive and significant, the interaction terms are not significant. Finally, no comparative effects, as measured by the effect of the regional median of equivalent income, emerge from our model.

Moderation and inclusiveness in excess of the general reference level also have a positive marginal impact on happiness, with coefficients of 3.92 and 5.06 respectively. Where inclusiveness is concerned, this is consistent with studies that show how higher levels of trust and of general openness towards other people are associated with greater happiness. In the case of moderation, the result might reflect a comfortable distance from events, inducing an ability to filter them into feelings in a more level-headed manner. The intuition is particularly suitable to our sample, entirely composed of European countries where disastrous phenomena such as famine, epidemics, war on domestic territory, and destruction wrought by extreme weather conditions are virtually unknown. While a strong disposition towards moderation might not be able to curb unhappiness from such occurrences, it can reduce the impact of run-of-the-mill unpleasant happenings. This explanation is, however, entirely speculative.

Finally, results on the set of controls are entirely aligned with previous

literature. Among the circumstances positively associated with happiness we find good health, marriage or cohabitation, residence in a safe neighbourhood, intense religious belief, and the enjoyment of close friendships. On the other hand, belonging to a discriminated group, having limited social interaction, and living in a large city exert a negative impact.

We have so far looked at moderation and inclusiveness as separate dimensions. In the spirit of understanding whether the interaction of the two might impact on happiness directly, and also as a manner of carrying out sensitivity analysis, we estimated two further specifications of our regression: the first is based on the simple consideration of the Cartesian quadrants defined by the two orthogonal factors, the second is founded on non-hierarchical *k-means* cluster analysis [Anderberg, 1973; Everitt, 1980].

Table A.5 shows that significant inequality aversion is found in the two high-moderation quadrants; high inclusiveness intensifies the phenomenon slightly, but it is not a requisite for its existence. The coefficient for interaction between inequality and the high-inclusiveness, low-moderation quadrant dummy is not significant, although it has the expected negative sign.

Cluster analysis reveals the presence of four clusters, described in Table A.6. Regressions that include cluster dummies (Table A.7) give insights similar to those presented above: significant inequality aversion is found for Cluster 2, which comprises people with high moderation scores, while a non-significant negative coefficient is found for the highly inclusive individuals in Cluster 4.

In general, when below-average moderation is accompanied by aboveaverage inclusiveness, the impact of the former appears to outweigh the impact of the latter, consistently with the results presented above. It is also possible that intra-class variability affects the estimates more when it comes to inclusiveness than when it comes to moderation, notwithstanding similar distributions and independent of whether standardization procedures are employed to neutralize outliers. This suggests that the effect of relative inclusiveness goes in the same direction but is both weaker and more markedly non-linear than the effect of relative moderation.

As a further exercise of sensitivity analysis, we included country dummies in the estimation of 2, to control for idiosyncratic effects; in order to do that, we need to take out the national median of incomes, lest we incur perfect collinearity. The results on inequality and values are stable, and only about half of the dummies are significant. We favour the specification that includes the national median of income because, although potentially less comprehensive in meaning than a country dummy, it measures a dimension that is clearly understandable, and its effect can be easily estimated in interaction with values.

6 Conclusions

This paper set out to understand whether it is possible to model the relationship between inequality and happiness in a way that is consistently appropriate across domains and takes into account both positional and interpretational effects. We looked at the heterogeneity of values, inclinations and beliefs as a possible unifying explanation for the existence of different reactions to inequality.

Drawing on data for twenty-three countries from the second round of the European Social Survey, carried out in 2004, we found that individual views on a wide range of themes can be effectively summarized by two orthogonal dimensions: moderation and inclusiveness. The former is defined as a tendency to take mild stands on issues rather than extreme ones; the latter is defined as the degree of support for a social model that grants equal rights to everyone who willingly subscribes to a shared set of rules, regardless of background and circumstances.

We ran a set of ordered logits where the degree of happiness on a 0-10 scale appears on the left-hand side, while distributive indicators interacted with individual deviations from country medians in terms of moderation and inclusiveness appear on the right-hand side, supplemented by a set of standard controls. We chose to look at deviations from country medians rather than raw levels in order to account for the fact that all the countries in the sample are democracies, where redistributive policies are endogenous with respect to values insofar as they capture, at least to some extent, the opinions and desires of the representative voter.

Values turn out to matter when it comes to determining the sign and the intensity of the relationship between inequality and happiness. In particular, individuals who are either more moderate or more inclusive than their average compatriots tend to prefer lower levels of inequality. In the case of moderation, inequality aversion can be read in terms of a desire for stability: people who are reluctant to take strong stands probably also resent social tension and unrest, which often accompany inequality. In the case of inclusiveness, the main element at play is likely to be a negative reaction to perceived unfairness. The effect of moderation appears to be stronger than the effect of inclusiveness: instrumental inequality aversion is more frequent and more intense than substantive inequality aversion.

Worldview effects appeared to be significant also with reference to the impact of personal income on happiness. The net effect of the logarithm of own equivalent income is positive for all clusters, but it is remarkably less intense, approximating zero, for people with a strong drive towards inclusion.

The marginal effect of values was found to be positive and significant for those who exceed either average moderation or average inclusiveness. While the result on inclusiveness is expected, on account of a well-known relationship between openness towards others, trust and happiness, the positive sign on moderation was somewhat unanticipated. One possible explanation may start with the observation that the sample only includes developed countries, largely immune from disastrous phenomena such as widespread extreme poverty or war on domestic territory. Most events that can be perceived as unpleasant, barring health conditions that are controlled for in our regressions, are probably minor; if moderation is associated with a comfortable emotional distance from mundane disruptions, then it is bound to have a positive impact on happiness.

References

- ALESINA, A., R. DI TELLA, AND R. MCCULLOCH (2004): "Inequality and happiness: are Europeans and Americans different?," *Journal of Public Economics*, 88, 2009–2042.
- ALESINA, A., AND R. PEROTTI (1996): "Income distribution, political instability, and investment," *European Economic Review*, 40, 1203–1228.
- ANDERBERG, M. R. (1973): *Cluster analysis for applications*. Academic Press, New York.
- BANERJEE, A., AND E. DUFLO (2003): "Inequality and growth: what can the data say?," *Journal of Economic Growth*, 8, 267–299.

- BECCHETTI, L., S. CASTRIOTA, AND O. GIUNTELLA (2006): "The effects of age and job protection on the welfare costs of inflation and unemployment: a source of ECB anti-inflation bias?," CEIS Working Paper 245.
- BENZÉCRI, J.-P. (1973): L'Analyse des donnees: l'analyse des correspondances. Dunod, Paris.
- BERTOLA, G. (1999): "Income distribution and macroeconomics," in Handbook of income distribution, ed. by A. B. Atkinson, and F. Bourguignon. North-Holland, Amsterdam.
- BÉNABOU, R., AND J. TIROLE (2006): "Belief in a just world and redistributive politics," *Quarterly Journal of Economics*, 121, 699–746.
- BROSNAN, S. F., AND F. B. M. DEWAAL (2003): "Monkeys reject unequal pay," *Nature*, 425, 297–299.
- CHOSSUDOVSKY, M. (1972): "Optimal policy configurations under alternative community preferences," *Kyklos*, 25, 754–770.
- CLARK, A. (2003): "Inequality aversion and income mobility: a direct test," DELTA Working Paper 11.
- CLARK, A., AND A. OSWALD (1996): "Satisfaction and comparison income," *Journal of Public Economics*, 61, 359–381.
- COHEN, G. A. (1989): "On the currency of egalitarian justice," *Ethics*, 99, 906–944.
- DI TELLA, R., AND R. MCCULLOCH (2006): "Some uses of happiness data in economics," *Journal of Economic Perspectives*, 20, 25–46.
- DWORKIN, R. (1981): "What is equality? Part 1: Equality of welfare," *Philosophy and Public Affairs*, 10, 185–246.
- EASTERLIN, R. A. (1974): "Does economic growth improve the human lot? Some empirical evidence," in *Nations and households in economic* growth: essays in honor of Moses Abramowitz, ed. by P. A. David, and M. W. Reder. Academic Press, New York.
 - (2006): "A brief history of quality-of-life studies in economics," in *The quality of life research movement: past, present and future*, ed. by M. J. S. et al. Springer, Dordrecht.

EVERITT, B. S. (1980): *Cluster analysis*. Heineman Educational Books, London.

- FEHR, E., AND U. FISCHBACHER (2002): "Why social preferences matter: the impact of non-selfish motives on competition, cooperation and incentives," *Economic Journal*, 112, C1–C33.
- FEHR, E., AND K. M. SCHMIDT (1999): "A theory of fairness, competition, and cooperation," *Quarterly Journal of Economics*, 3, 817–868.
- FERNÁNDEZ, R., AND A. FOGLI (2006): "Fertility: the role of culture and family experience," Journal of the European Economic Association, 4, 552–561.
- FORBES, K. J. (2000): "A reassessment of the relationship between inequality and growth," American Economic Review, 90, 869–887.
- FREY, B., AND A. STUTZER (2002): "What can economists learn from happiness research?," Journal of Economic Literature, 40, 402–435.
- (2005): "Happiness research: state and prospects," *Review of Social Economy*, 62, 207–228.
- GRAHAM, C., AND A. FELTON (2006): "Inequality and happiness: insights from Latin America," *Journal of Economic Inequality*, 4, 107–122.
- GUISO, L., P. SAPIENZA, AND L. ZINGALES (2003): "People's opium? Religion and economic attitudes," *Journal of Monetary Economics*, 50, 225– 282.
- GUISO, L., P. SAPIENZA, AND L. ZINGALES (2006): "Does culture affect economic outcomes?," NBER Working Paper 11999.
- HENRICH, J., R. BOYD, S. BOWLES, C. CAMERER, E. FEHR, AND H. GIN-TIS (2004): Foundations of human sociality: economic experiments and ethnographic evidence from fifteen small-scale societies. Oxford Unviersity Press, Oxford.
- HIRSCH, F. (1976): *The social limits to growth.* Routledge and Kegan Paul, London.

- HOFFMAN, E., K. MCCABE, K. SHACHAT, AND V. SMITH (1994): "Preferences, property rights, and anonymity in bargaining games," *Games and Economic Behavior*, 7, 346–380.
- HOFFMAN, E., K. MCCABE, AND V. L. SMITH (1996): "On expectations and the monetary stakes in Ultimatum Games," *International Journal of Game Theory*, 25, 289–301.
- HORII, T., Y. JIN, AND R. LEVITT (2005): "Modeling and analyzing cultural influences on project team performance," *Computational and Mathematical Organization Theory*, 10, 305–321.
- INGLEHART, R., AND W. BAKER (2000): "Modernization, cultural change and the persistence of traditional values," *American Sociological Review*, 65, 19–51.
- KAHNEMAN, D., AND A. B. KRUGER (2006): "Developments in the measurement of subjective well-being," *Journal of Economic Perspectives*, 20, 3–24.
- KEEFER, P., AND S. KNACK (2002): "Polarization, politics and property rights: links between inequality and growth," *Public Choice*, 111, 127–154.
- KOHLER, H.-P., J. R. BEHRMAN, AND A. SKYTTHE (2005): "Partner + children = happiness? The effects of partnerships and fertility on wellbeing," *Population and Development Review*, 31, 407–445.
- LEBART, L., A. MORINEAU, AND K. WARWICK (1984): Multivariate descripitive statistical analysis: correspondence analysis and related techniques for large matrices. John Wiley and Sons, New York.
- LEFRANC, A., N. PISTOLESI, AND A. TRANNOY (2006): "Equality of opportunity: Definitions and testable conditions, with an application to income in France," ECINEQ Working Paper 53.
- LUTTMER, E. F. P. (2005): "Neighbors as negatives: relative earnings and well-being," *Quarterly Journal of Economics*, 120, 963–1002.
- MCFARLIN, D., AND R. W. RICE (1992): "The role of facet importance as a moderator in job satisfaction process," *Journal of Organizational Behavior*, 13, 41–54.

- MELTZER, A., AND S. RICHARD (1981): "A rational theory of the size of government," *Journal of Political Economy*, 89, 914–927.
- QUAH, D. (2003): "One third of the world's growth and inequality," in Inequality and growth: theory and policy implications, ed. by T. Eicher, and S. J. Turnovsky. MIT Press, Cambridge, MA.
- RAWLS, J. (1971): A theory of justice. The Belknap Press of Harvard University Press, Cambridge, Massachusetts.
- ROEMER, J. E. (2006): "The 2006 World Development Report: equity and development," *Journal of Economic Inequality*, 4, 233–244.
- ROJAS, M. (2007): "Heterogeneity in the relationship between income and happiness: a conceptual-referent-theory explanation," *Journal of Economic Psychology*, 28, 1–14.
- SAGIV, L., AND S. SCHWARTZ (2000): "Value priorities and subjective wellbeing: direct relations and congruity effects," *European Journal of Social Psychology*, 30, 177–198.
- STUTZER, A. (2004): "The role of income aspirations in individual happiness," *Journal of Economic Behavior and Organization*, 54, 89–109.
- TABELLINI, G. (2006): "Culture and institutions: economic development in the regions of Europe," IGIER Working Paper.
- VEBLEN, T. (1899): The theory of the leisure class. made available online by The Gutenberg Project, http://www.gutenberg.org/etext/833.
- WORLD BANK (2005): World Development Report 2006: Equity and Development. World Bank, Washington.

A Appendix

| Country | Starting | Deleted of | oservations | Imputed incomes | | |
|--|------------|------------|-------------|-----------------|----------|--|
| J. J | size | N | Fraction | Ν | Fraction | |
| Austria | 2,256 | 654 | 0.29 | 627 | 0.28 | |
| Belgium | 1,778 | 142 | 0.08 | 343 | 0.19 | |
| Czech Republic | 3,026 | 1245 | 0.41 | 473 | 0.16 | |
| Denmark | 1,487 | 239 | 0.16 | 127 | 0.09 | |
| Estonia | 1,989 | 700 | 0.35 | 232 | 0.12 | |
| Finland | 2,022 | 135 | 0.07 | 121 | 0.06 | |
| France ⁺ | 1,806 | 402 | 0.22 | 0 | 0.00 | |
| Germany | 2,870 | 440 | 0.15 | 480 | 0.17 | |
| Greece | 2,406 | 358 | 0.15 | 630 | 0.26 | |
| Hungary | 1,498 | 345 | 0.23 | 131 | 0.09 | |
| Iceland | 579 | 72 | 0.12 | 57 | 0.10 | |
| Ireland | 2,286 | 483 | 0.21 | 379 | 0.17 | |
| Luxembourg | 1,635 | 340 | 0.21 | 457 | 0.28 | |
| Netherlands | 1,881 | 136 | 0.07 | 197 | 0.10 | |
| Norway | 1,760 | 50 | 0.03 | 39 | 0.02 | |
| Poland | 1,716 | 409 | 0.24 | 198 | 0.12 | |
| Portugal | 2,052 | 426 | 0.21 | 678 | 0.33 | |
| Slovakia | 1,512 | 490 | 0.32 | 325 | 0.21 | |
| Slovenia | 1,442 | 290 | 0.20 | 208 | 0.14 | |
| Spain | 1,663 | 342 | 0.21 | 446 | 0.27 | |
| Sweden | 1,948 | 194 | 0.10 | 98 | 0.05 | |
| Switzerland | 2,141 | 281 | 0.13 | 348 | 0.16 | |
| United Kingdom | 1,897 | 142 | 0.07 | 337 | 0.18 | |
| Total | $43,\!650$ | 8,315 | 0.19 | 6,931 | 0.16 | |

| Table A.1: Deleted observations and imputed incomes, by cou | intry |
|---|-------|
|---|-------|

+ There are no imputed incomes for France on account of the unavailability of subjective quality-of-life measures. All observations lacking income information were therefore deleted.

Table A.2: Elementary value input for multiple correspondence analysis

| Thematic area | ESS variable | Question text | Response scale |
|------------------|--------------|---|-------------------|
| | PPLTRST | Generally speaking, would you say that most people can be trusted, or that you can't be too careful? | 11-point |
| Trust | WRYTRDH | How worried are you of being treated dishonestly? | 4-point |
| | BSNPRFT | Businesses are only interested in making profits and not in improving service or quality (agreement) | 5-point |
| | CTZHLPO | Citizens should spend at least some of their free time helping others (agreement) | 5-point |
| Solidarity | SCBEVTS | Society would be better off it everyone just looked after themselves (agreement) | 5-point |
| | GINCDIF | The government should take measures to reduce differences in income levels (agreement) | 5-point |
| | SLCNFLW | How wrong: someone selling something second-hand and concealing some or all of its faults? | 4-point |
| Compliance | PBOFVRW | How wrong: a public official asking someone for a favour or bribe in return for their services? | 4-point |
| with law | CTZCHTX | Citizens should not cheat on their taxes (agreement) | 5-point |
| | POLINTR | How interested would you say you are in politics? | 4-point |
| Civic | VOTE | Did you vote in the last national election? | YES/NO/NE |
| engagement | NWSPTOT | On an average weekday, how much time, in total, do you spend reading newspapers? | 8-point |
| | WMCPWRK | A woman should be prepared to cut down on her paid work for the sake of her family (agreement) | 5-point |
| Family and | MNRSPHM | Men should take as much responsibility as women for the home and children (agreement) | 5-point |
| genaer roles | MNRGTJB | When jobs are scarce, men should have more right to a job than woman (agreement) | 5-point |
| | FREEHMS | Gay men and lesbians should be free to live their own life as they wish (agreement) | 5-point |
| Minorities | IMWBCNT | Is [cntry] made a worse or a better place to live by people coming to live here from other countries? | 11-point |
| | IMDFETN | To what extend do you think [cntry] should allow people of a different race to come and live here? | 11-point |

| Var | Response | Dim1 | Dim2 | Contr1 | Contr2 | Quality | Mass | Inertia |
|-------------|----------------------------|---------|---------|--------|--------|---------|--------|---------|
| | You can't be too careful | -1.0472 | -0.6244 | 0.0200 | 0.0082 | 0.0889 | 0.0030 | 0.0118 |
| | 1 | -0.7413 | -0.6472 | 0.0069 | 0.0061 | 0.0371 | 0.0020 | 0.0120 |
| | 2 | -0.3746 | -0.5753 | 0.0034 | 0.0093 | 0.0363 | 0.0040 | 0.0116 |
| | 3 | -0.0504 | -0.4469 | 0.0001 | 0.0088 | 0.0258 | 0.0062 | 0.0111 |
| E | 4 | 0.0684 | -0.1335 | 0.0002 | 0.0008 | 0.0091 | 0.0064 | 0.0111 |
| ΓR. | 5 | 0.0649 | -0.0901 | 0.0003 | 0.0007 | 0.0118 | 0.0125 | 0.0097 |
| PLC | 6 | 0.3329 | 0.2008 | 0.0042 | 0.0018 | 0.0191 | 0.0062 | 0.0111 |
| Ч | 7 | 0.3465 | 0.4755 | 0.0059 | 0.0129 | 0.0585 | 0.0080 | 0.0107 |
| | 8 | 0.1474 | 0.7028 | 0.0007 | 0.0187 | 0.0618 | 0.0053 | 0.0113 |
| | 9 | -0.0292 | 0.8783 | 0.0000 | 0.0063 | 0.0227 | 0.0012 | 0.0122 |
| | Most people can be trusted | -0.5470 | 0.5488 | 0.0014 | 0.0016 | 0.0103 | 0.0007 | 0.0122 |
| Ξ | Not at all worried | -0.0698 | 0.1976 | 0.0001 | 0.0010 | 0.0175 | 0.0148 | 0.0092 |
| D. | A bit worried | 0.1844 | 0.1510 | 0.0056 | 0.0041 | 0.0170 | 0.0140 | 0.0052 |
| IT, | Fairly worried | 0.1044 | 0.0043 | 0.0030 | 0.0000 | 0.0439 | 0.0271 | 0.0004 |
| VR | Variation Variation | -0.1203 | -0.2282 | 0.0011 | 0.0040 | 0.0179 | 0.0108 | 0.0101 |
| > | Very worried | -0.9108 | -0.2029 | 0.0145 | 0.0008 | 0.0371 | 0.0028 | 0.0119 |
| F | Disagree strongly | -0.8050 | 0.5514 | 0.0035 | 0.0019 | 0.0159 | 0.0009 | 0.0123 |
| RF | Disagree | 0.2462 | 0.3586 | 0.0029 | 0.0072 | 0.0402 | 0.0079 | 0.0107 |
| NP | Neither agree nor disagree | 0.3516 | 0.1400 | 0.0077 | 0.0014 | 0.0692 | 0.0102 | 0.0102 |
| $_{\rm BS}$ | Agree | 0.2379 | -0.1426 | 0.0084 | 0.0035 | 0.0668 | 0.0243 | 0.0070 |
| | Agree strongly | -0.8574 | -0.1037 | 0.0554 | 0.0009 | 0.2129 | 0.0123 | 0.0097 |
| 0 | Disagree strongly | -1.8246 | -0.5613 | 0.0075 | 0.0008 | 0.0253 | 0.0004 | 0.0124 |
| CPO | Disagree | -0.1412 | -0.2036 | 0.0003 | 0.0008 | 0.0402 | 0.0079 | 0.0107 |
| [H] | Neither agree nor disagree | 0.2652 | -0.0840 | 0.0046 | 0.0005 | 0.1277 | 0.0106 | 0.0101 |
| E | Agree | 0.2099 | -0.0637 | 0.0089 | 0.0010 | 0.1152 | 0.0330 | 0.0051 |
| | Agree strongly | -0.9893 | 0.4282 | 0.0525 | 0.0114 | 0.2289 | 0.0088 | 0.0105 |
| 10 | Disagree strongly | -0.4747 | 0.7168 | 0.0158 | 0.0418 | 0.2050 | 0.0115 | 0.0099 |
| Ĩ.L.A | Disagree | 0.3744 | 0.0197 | 0.0224 | 0.0001 | 0.1385 | 0.0261 | 0.0066 |
| Ē | Neither agree nor disagree | 0.1633 | -0.3549 | 0.0013 | 0.0069 | 0.1202 | 0.0077 | 0.0108 |
| SCI | Agree | -0.1897 | -0.6238 | 0.0017 | 0.0207 | 0.0690 | 0.0075 | 0.0108 |
| | Agree strongly | -1.5097 | -0.4811 | 0.0384 | 0.0045 | 0.1354 | 0.0028 | 0.0119 |
| | Disagree strongly | -0.4795 | 0.4171 | 0.0022 | 0.0019 | 0.0143 | 0.0016 | 0.0122 |
| OIF | Disagree | 0.3883 | 0.2069 | 0.0067 | 0.0022 | 0.0462 | 0.0072 | 0.0109 |
| ICI | Neither agree nor disagree | 0.3859 | 0.0336 | 0.0078 | 0.0001 | 0.0657 | 0.0086 | 0.0106 |
| 8 | Agree | 0.2717 | -0.1119 | 0.0106 | 0.0021 | 0.0670 | 0.0235 | 0.0072 |
| - | Agree strongly | -0.8008 | 0.0132 | 0.0575 | 0.0000 | 0.2306 | 0.0147 | 0.0092 |
| 3 | Not wrong at all | -1.2506 | -0.7478 | 0.0056 | 0.0023 | 0.0694 | 0.0006 | 0.0124 |
| ΕĽ | A bit wrong | 0.0369 | -0.3399 | 0.0000 | 0.0031 | 0.0948 | 0.0038 | 0.0116 |
| CN | Wrong | 0.2431 | -0.1340 | 0.0094 | 0.0033 | 0.0841 | 0.0260 | 0.0067 |
| SL | Seriously wrong | -0.2266 | 0.2063 | 0.0079 | 0.0076 | 0.1817 | 0.0252 | 0.0068 |
| 2 | Not wrong at all | -0.9787 | -0.6764 | 0.0032 | 0.0018 | 0.0791 | 0.0005 | 0.0124 |
| ∕.B∖ | A bit wrong | 0.0886 | -0.5233 | 0.0001 | 0.0033 | 0.0722 | 0.0017 | 0.0121 |
| PFI OFI | Wrong | 0 1646 | -0.3626 | 0.0025 | 0.0139 | 0.0935 | 0.0149 | 0.0092 |
| BC | Seriously wrong | -0.0536 | 0.1725 | 0.0007 | 0.0081 | 0.1818 | 0.0385 | 0.0038 |
| | Disagroo strongly | 1 0103 | 0.1576 | 0.0073 | 0.0002 | 0.1010 | 0.0000 | 0.0000 |
| X | Disagree strongly | 0 1182 | 0.1570 | 0.0073 | 0.0002 | 0.0227 | 0.0012 | 0.0122 |
| E H | Noither agree per diaggree | 0.1516 | -0.1050 | 0.0003 | 0.0001 | 0.0003 | 0.0033 | 0.0117 |
| Z | A men | 0.1310 | -0.0422 | 0.0010 | 0.0001 | 0.1012 | 0.0073 | 0.0108 |
| 5 | Agree | 0.3489 | -0.1155 | 0.0220 | 0.0027 | 0.1090 | 0.0295 | 0.0039 |
| | Net at all interests 1 | -0.0997 | 0.2000 | 0.0420 | 0.0000 | 0.4119 | 0.0141 | 0.0093 |
| ΤB | Not at all interested | -0.0000 | -0.0617 | 0.0176 | 0.0240 | 0.1637 | 0.0093 | 0.0104 |
| IIN | Hardly interested | 0.0690 | -0.2231 | 0.0006 | 0.0072 | 0.0646 | 0.0203 | 0.0079 |
| 10, | Quite interested | 0.2306 | 0.2756 | 0.0065 | 0.0108 | 0.1059 | 0.0201 | 0.0080 |
| ц | Very interested | -0.1455 | 0.7950 | 0.0008 | 0.0260 | 0.1193 | 0.0058 | 0.0112 |
| E | No | -0.2534 | -0.4323 | 0.0044 | 0.0149 | 0.1104 | 0.0112 | 0.0100 |
| ΓO^7 | Yes | 0.0562 | 0.1087 | 0.0008 | 0.0033 | 0.1940 | 0.0399 | 0.0035 |
| > | Not eligible to vote | 0.1353 | 0.1165 | 0.0005 | 0.0004 | 0.1148 | 0.0044 | 0.0115 |

Table A.3: Item factor loadings and goodness-of-fit statistics

| Var | Response | Dim1 | Dim2 | Contr1 | Contr2 | Quality | Mass | Inertia |
|-------------|----------------------------|---------|---------|--------|--------|---------|--------|---------|
| | No time at all | -0.3160 | -0.2320 | 0.0093 | 0.0058 | 0.0812 | 0.0152 | 0.0091 |
| | Less than 0.5 hour | 0.0942 | 0.0590 | 0.0009 | 0.0004 | 0.0370 | 0.0174 | 0.0086 |
| TC | 0.5 to 1 hr | 0.1648 | 0.1330 | 0.0026 | 0.0020 | 0.0663 | 0.0158 | 0.0090 |
| Ĕ | 1 to 1.5 hr | 0.1326 | 0.0888 | 0.0004 | 0.0002 | 0.0094 | 0.0042 | 0.0116 |
| MS | 1.5 to 2 hr | 0.0398 | -0.0503 | 0.0000 | 0.0000 | 0.0040 | 0.0016 | 0.0121 |
| MN | 2 to 2.5 hr | 0.1277 | 0.0559 | 0.0001 | 0.0000 | 0.0046 | 0.0006 | 0.0124 |
| | 2.5 to 3 hr | -0.1789 | 0.1822 | 0.0001 | 0.0001 | 0.0052 | 0.0003 | 0.0124 |
| | More than 3 hr | -0.1673 | 0.0554 | 0.0001 | 0.0000 | 0.0047 | 0.0005 | 0.0124 |
| K | Disagree strongly | -0.8380 | 1.2236 | 0.0180 | 0.0445 | 0.2253 | 0.0042 | 0.0116 |
| WB | Disagree | 0.2911 | 0.3848 | 0.0062 | 0.0125 | 0.0636 | 0.0119 | 0.0098 |
| E S | Neither agree nor disagree | 0.3271 | 0.0005 | 0.0082 | 0.0000 | 0.0991 | 0.0125 | 0.0097 |
| M | Agree | 0.1859 | -0.3655 | 0.0044 | 0.0196 | 0.1814 | 0.0207 | 0.0078 |
| 5 | Agree strongly | -1.2744 | -0.3475 | 0.0615 | 0.0053 | 0.2263 | 0.0062 | 0.0111 |
| Z | Disagree strongly | -1.6401 | -0.6477 | 0.0070 | 0.0013 | 0.0242 | 0.0004 | 0.0124 |
| IHd | Disagree | -0.2740 | -0.7251 | 0.0010 | 0.0083 | 0.0332 | 0.0022 | 0.0120 |
| RSI | Neither agree nor disagree | 0.1796 | -0.4752 | 0.0009 | 0.0071 | 0.0427 | 0.0044 | 0.0115 |
| ĮN. | Agree | 0.4558 | -0.1885 | 0.0368 | 0.0073 | 0.2974 | 0.0290 | 0.0060 |
| 4 | Agree strongly | -0.6524 | 0.4855 | 0.0506 | 0.0326 | 0.3767 | 0.0195 | 0.0081 |
| В | Disagree strongly | -0.5052 | 1.0501 | 0.0179 | 0.0900 | 0.4080 | 0.0115 | 0.0099 |
| ſĿ | Disagree | 0.4894 | 0.1273 | 0.0272 | 0.0021 | 0.1416 | 0.0186 | 0.0083 |
| RG | Neither agree nor disagree | 0.3018 | -0.3207 | 0.0059 | 0.0077 | 0.0772 | 0.0106 | 0.0101 |
| ΛĮΝ | Agree | 0.0171 | -0.7548 | 0.0000 | 0.0426 | 0.1760 | 0.0106 | 0.0101 |
| 4 | Agree strongly | -1.5151 | -0.7071 | 0.0614 | 0.0155 | 0.2503 | 0.0044 | 0.0115 |
| ŝ | Disagree strongly | -1.2129 | -0.7597 | 0.0284 | 0.0129 | 0.1406 | 0.0032 | 0.0118 |
| MH | Disagree | -0.0919 | -0.7940 | 0.0003 | 0.0217 | 0.0759 | 0.0049 | 0.0114 |
| EEI | Neither agree nor disagree | 0.0906 | -0.5395 | 0.0004 | 0.0153 | 0.0571 | 0.0074 | 0.0108 |
| FRJ | Agree | 0.4130 | -0.0809 | 0.0241 | 0.0011 | 0.1559 | 0.0231 | 0.0073 |
| | Agree strongly | -0.3482 | 0.7124 | 0.0126 | 0.0613 | 0.3391 | 0.0170 | 0.0087 |
| | Country made worse | -1.3786 | -0.9029 | 0.0066 | 0.0134 | 0.0566 | 0.0023 | 0.0120 |
| | 1 | -0.6843 | -0.9044 | 0.0304 | 0.0151 | 0.1353 | 0.0026 | 0.0119 |
| | 2 | -0.3571 | -0.7839 | 0.0031 | 0.0175 | 0.0607 | 0.0040 | 0.0116 |
| н | 3 | -0.0137 | -0.5541 | 0.0000 | 0.0126 | 0.0401 | 0.0058 | 0.0112 |
| ON CONTRACT | 4 | 0.2465 | -0.2781 | 0.0022 | 0.0033 | 0.0167 | 0.0060 | 0.0112 |
| VB | 5 | 0.1702 | 0.0958 | 0.0031 | 0.0011 | 0.0406 | 0.0173 | 0.0086 |
| M | 6 | 0.3740 | 0.2714 | 0.0049 | 0.0030 | 0.0247 | 0.0058 | 0.0112 |
| | 7 | 0.2782 | 0.5574 | 0.0026 | 0.0119 | 0.0423 | 0.0054 | 0.0113 |
| | 8 | -0.0119 | 0.8448 | 0.0000 | 0.0204 | 0.0576 | 0.0040 | 0.0116 |
| | 9 | -0.2717 | 1.1180 | 0.0005 | 0.0101 | 0.0285 | 0.0011 | 0.0122 |
| | Country made better | -0.8320 | 1.2718 | 0.0052 | 0.0141 | 0.0522 | 0.0012 | 0.0122 |
| NT | Allow none | -0.8013 | -0.9506 | 0.0284 | 0.0464 | 0.2337 | 0.0072 | 0.0109 |
| Ē | Allow a few | 0.0142 | -0.3260 | 0.0000 | 0.0144 | 0.0575 | 0.0191 | 0.0082 |
| QW | Allow some | 0.2946 | 0.2731 | 0.0120 | 0.0120 | 0.1135 | 0.0226 | 0.0074 |
| I | Allow many | -0.1737 | 1.0522 | 0.0012 | 0.0516 | 0.1540 | 0.0066 | 0.0110 |

 $\label{eq:addings} {\rm Table \ A.3: \ Item \ factor \ loadings \ and \ goodness-of-fit \ statistics \ (cont.)}$

 Table A.4: Ordered logit estimates for happiness: benchmark

| | Estimate | $\mathbf{Std} \ \mathbf{Err}^+$ | Pr>ChiSq |
|---|---------------|---------------------------------|-------------|
| Equivalent income | | | • |
| Log own | 0.175 | 0.031 | 0.000 |
| Std interquartile range, regional | 0.198 | 0.190 | 0.297 |
| Log median, national | 0.225 | 0.164 | 0.169 |
| Log median, regional | 0.095 | 0.136 | 0.484 |
| Demographics | | | |
| Gender: female | 0.187 | 0.036 | 0.000 |
| Age | -0.070 | 0.007 | 0.000 |
| Age squared | 0.001 | 0.000 | 0.000 |
| Self-reported health (baseline | = very good |) | • |
| Good | -0.574 | 0.042 | 0.000 |
| Fair | -1.122 | 0.103 | 0.000 |
| Bad | -1.860 | 0.125 | 0.000 |
| Very bad | -2.527 | 0.366 | 0.000 |
| Marital status (baseline $=$ max | rried or in r | egistered col | nabitation) |
| Separated | -0.893 | 0.162 | 0.000 |
| Divorced | -0.581 | 0.053 | 0.000 |
| Widowed | -0.960 | 0.102 | 0.000 |
| Never married | -0.565 | 0.071 | 0.000 |
| Children | | | |
| Children living at home | -0.052 | 0.058 | 0.372 |
| Children living outside the home | 0.146 | 0.028 | 0.000 |
| Social ties | - | | - |
| At least one close friend | 0.622 | 0.060 | 0.000 |
| Frequency of social activity | 0.123 | 0.007 | 0.000 |
| Location (baseline $=$ city cent | er) | | |
| Suburbs or outskirts of big city | -0.021 | 0.052 | 0.688 |
| Town or small city | 0.016 | 0.073 | 0.828 |
| Country village | 0.048 | 0.070 | 0.494 |
| Farm or home in countryside | 0.179 | 0.095 | 0.059 |
| Feeling of safety in own neight | ourhood (b | aseline = ve | ry safe) |
| Safe | -0.181 | 0.044 | 0.000 |
| Unsafe | -0.332 | 0.036 | 0.000 |
| Very unsafe | -0.472 | 0.144 | 0.001 |
| Job status (baseline $=$ employ | ee) | | |
| Student | 0.202 | 0.089 | 0.022 |
| Unemployed, looking for job | -0.584 | 0.125 | 0.000 |
| Unemployed, not looking | -0.442 | 0.143 | 0.002 |
| Permanently sick or disabled | 0.105 | 0.176 | 0.550 |
| Retired | 0.136 | 0.073 | 0.064 |
| Community or military service | 0.353 | 0.315 | 0.263 |
| Housework | 0.052 | 0.052 | 0.316 |
| Other | -0.108 | 0.211 | 0.609 |
| Other factors | | | |
| Years in formal education | 0.017 | 0.009 | 0.047 |
| Intensity of religious belief | 0.051 | 0.009 | 0.000 |
| Belongs to discriminated group | -0.484 | 0.118 | 0.000 |
| Homeowner | 0.132 | 0.052 | 0.011 |
| Model fit statistics: | 1 | 0.000 | |
| Prob > Chi Square (Wald) $D = 1 D^2$ | | 0.000 | |
| rseudo K | 1 | 0.048 | |

+Standard errors are adjusted for clustering at the regional level

Table A.5: Ordered logit estimates for happiness: quadrants

| | Estimate | $\mathbf{Std} \ \mathbf{Err}^+$ | $\mathbf{Pr} > \mathbf{ChiSq}$ | | | |
|---|------------|---------------------------------|--------------------------------|--|--|--|
| Equivalent income | | | | | | |
| (baseline for interaction terms: SW Quadrant, | Mod < medi | an; Incl < me | edian) | | | |
| Log own | 0.283 | 0.064 | 0.000 | | | |
| *NW Quadrant Mod < median; Incl > median | -0.163 | 0.086 | 0.057 | | | |
| *NE Quadrant Mod > median; Incl > median | -0.227 | 0.079 | 0.004 | | | |
| *SE Quadrant $Mod > median; Incl < median$ | -0.117 | 0.062 | 0.057 | | | |
| Std interquartile range, regional | 0.740 | 0.279 | 0.008 | | | |
| *NW Quadrant | -0.293 | 0.298 | 0.325 | | | |
| *NE Quadrant | -0.883 | 0.307 | 0.004 | | | |
| *SE Quadrant | -0.640 | 0.265 | 0.016 | | | |
| Log median, national | 0.732 | 0.306 | 0.017 | | | |
| *NW Quadrant | -0.417 | 0.348 | 0.230 | | | |
| *NE Quadrant | -0.302 | 0.282 | 0.284 | | | |
| *SE Quadrant | -0.310 | 0.277 | 0.264 | | | |
| Log median, regional | -0.278 | 0.324 | 0.391 | | | |
| *NW Quadrant | 0.300 | 0.394 | 0.446 | | | |
| *NE Quadrant | 0.097 | 0.295 | 0.742 | | | |
| *SE Quadrant | -0.160 | 0.278 | 0.566 | | | |
| Demographics | | | | | | |
| Gender: female | 0.134 | 0.037 | 0.000 | | | |
| Age | -0.062 | 0.009 | 0.000 | | | |
| Age squared | 0.006 | 0.000 | 0.000 | | | |
| Self-reported health (baseline – very good) | 0.000 | 0.000 | 0.000 | | | |
| Cood | 0.583 | 0.052 | 0.000 | | | |
| Fair | 1.077 | 0.075 | 0.000 | | | |
| Pad | 1 794 | 0.115 | 0.000 | | | |
| Dau Vow hod | -1.784 | 0.115 | 0.000 | | | |
| Manital status (baseline - mannied on in regist | -2.420 | 0.330 | 0.000 | | | |
| Generated | | 0.146 | 0.000 | | | |
| Diversed | -0.932 | 0.140 | 0.000 | | | |
| Divorced William I | -0.024 | 0.072 | 0.000 | | | |
| Widowed | -0.924 | 0.077 | 0.000 | | | |
| Never married | -0.031 | 0.049 | 0.000 | | | |
| | 0.007 | 0.040 | 0.160 | | | |
| Children living at home | -0.067 | 0.048 | 0.169 | | | |
| Children living outside the home | 0.144 | 0.051 | 0.004 | | | |
| Social ties | | | | | | |
| At least one close friend | 0.606 | 0.080 | 0.000 | | | |
| Frequency of social activity | 0.162 | 0.017 | 0.000 | | | |
| Location (baseline = city center) | | | | | | |
| Suburbs or outskirts of big city | -0.008 | 0.075 | 0.912 | | | |
| Town or small city | 0.051 | 0.061 | 0.408 | | | |
| Country village | 0.103 | 0.062 | 0.096 | | | |
| Farm or home in countryside | 0.249 | 0.107 | 0.020 | | | |
| Feeling of safety in own neighbourhood (baseline = very safe) | | | | | | |
| Safe | -0.107 | 0.045 | 0.016 | | | |
| Unsafe | -0.217 | 0.062 | 0.001 | | | |
| Very unsafe | -0.364 | 0.113 | 0.001 | | | |
| Job status | | | | | | |
| Student | 0.099 | 0.078 | 0.204 | | | |
| Unemployed, looking for job | -0.561 | 0.114 | 0.000 | | | |
| Unemployed, not looking | -0.380 | 0.148 | 0.010 | | | |
| Permanently sick or disabled | 0.085 | 0.140 | 0.610 | | | |
| Retired | 0.159 | 0.066 | 0.015 | | | |
| Community or military service | 0.295 | 0.266 | 0.268 | | | |
| Housework | 0.042 | 0.064 | 0.512 | | | |
| Other | -0.193 | 0.179 | 0.279 | | | |
| | | | | | | |

Table A.5: Ordered logit estimates for happiness: quadrants (cont.)

| | Estimate | Std Err ⁺ | \mathbf{Pr} >ChiSq | | |
|--|----------|----------------------|----------------------|--|--|
| Other factors | | | | | |
| Years in formal education | 0.004 | 0.006 | 0.424 | | |
| Intensity of religious belief | 0.054 | 0.009 | 0.000 | | |
| Belongs to discriminated group | -0.516 | 0.090 | 0.000 | | |
| Homeowner | 0.094 | 0.044 | 0.034 | | |
| Marginal effect of deviation from the median in values | | | | | |
| (baseline = SW Quadrant) | | | | | |
| NW Quadrant | 3.349 | 0.849 | 0.000 | | |
| NE Quadrant | 5.200 | 0.701 | 0.000 | | |
| SE Quadrant | 3.116 | 0.849 | 0.000 | | |
| Model fit statistics: | | | | | |
| Prob > Chi Square (Wald) | | 0.000 | | | |
| Pseudo \mathbb{R}^2 | | 0.059 | | | |

+ Standard errors are adjusted for clustering at the regional level.

Table A.6: Cluster means, standard deviations and descriptions

| | | Deviation from country median: | | | | |
|---------|------------|--------------------------------|--------|---------------|--------|----------------------|
| Cluster | Ν | Moderation | | Inclusiveness | | Description |
| | | Mean | St Dev | Mean | St Dev | |
| 1 | 12,500 | -0.268 | 0.335 | -0.032 | 0.205 | - Mod (weak) |
| 2 | $11,\!484$ | 0.360 | 0.137 | -0.076 | 0.191 | ++ Mod |
| 3 | 4,332 | -0.110 | 0.342 | -0.540 | 0.144 | - Mod (weak), – Incl |
| 4 | 7,019 | -0.062 | 0.296 | 0.511 | 0.209 | ++ Incl |

Table A.7: Ordered logit estimates for happiness: clusters

| | Estimate | $\mathbf{Std} \ \mathbf{Err}^+$ | Pr>ChiSq | | | |
|--|----------|---------------------------------|----------|--|--|--|
| Equivalent income (baseline for interaction terms = Cluster 1 $Mod < median$) | | | | | | |
| Log own | 0.219 | 0.049 | 0.000 | | | |
| *Cluster 2 $Mod > median$ | -0.068 | 0.057 | 0.238 | | | |
| *Cluster 3 Mod < median (weak); Incl < median | -0.058 | 0.075 | 0.436 | | | |
| *Cluster 4 $Incl > median$ | -0.156 | 0.075 | 0.039 | | | |
| Std interquartile range, regional | 0.584 | 0.201 | 0.004 | | | |
| *Cluster 2 | -0.734 | 0.197 | 0.000 | | | |
| *Cluster 3 | 0.261 | 0.336 | 0.439 | | | |
| *Cluster 4 | -0.652 | 0.289 | 0.024 | | | |
| Log median, national | 0.504 | 0.221 | 0.023 | | | |
| *Cluster 2 | -0.017 | 0.213 | 0.935 | | | |
| *Cluster 3 | 0.213 | 0.447 | 0.633 | | | |
| *Cluster 4 | -0.344 | 0.296 | 0.246 | | | |
| Log median, regional | -0.140 | 0.214 | 0.512 | | | |
| *Cluster 2 | -0.063 | 0.210 | 0.763 | | | |
| *Cluster 3 | -0.078 | 0.465 | 0.866 | | | |
| *Cluster 4 | 0.195 | 0.281 | 0.487 | | | |
| Demographics | | | | | | |
| Gender: female | 0.132 | 0.037 | 0.000 | | | |
| Age | -0.063 | 0.009 | 0.000 | | | |
| Age squared | 0.001 | 0.000 | 0.000 | | | |

| | Estimate | $\mathbf{Std} \ \mathbf{Err}^+$ | $\mathbf{Pr} > \mathbf{ChiSq}$ | | | |
|---|---------------|---------------------------------|--------------------------------|--|--|--|
| Self-reported health (baseline = very $good$) | | • | | | | |
| Good | -0.579 | 0.052 | 0.000 | | | |
| Fair | -1.070 | 0.074 | 0.000 | | | |
| Bad | -1.755 | 0.114 | 0.000 | | | |
| Very bad | -2.447 | 0.339 | 0.000 | | | |
| Marital status (baseline = married or in registered cohabitation) | | | | | | |
| Separated | -0.936 | 0.151 | 0.000 | | | |
| Divorced | -0.617 | 0.071 | 0.000 | | | |
| Widowed | -0.923 | 0.075 | 0.000 | | | |
| Never married | -0.634 | 0.048 | 0.000 | | | |
| Children | | | | | | |
| Children living at home | -0.061 | 0.048 | 0.196 | | | |
| Children living outside the home | 0.143 | 0.051 | 0.005 | | | |
| Social ties | | | | | | |
| At least one close friend | 0.615 | 0.080 | 0.000 | | | |
| Frequency of social activity | 0.162 | 0.017 | 0.000 | | | |
| Location (baseline = city center) | | | | | | |
| Suburbs or outskirts of big city | -0.018 | 0.074 | 0.806 | | | |
| *fino a qui Town or small city | 0.045 | 0.062 | 0.463 | | | |
| Country village | 0.093 | 0.063 | 0.143 | | | |
| Farm or home in countryside | 0.244 | 0.109 | 0.025 | | | |
| Feeling of safety in own neighbourhood (baseline | e = very safe | e) | | | | |
| Safe | -0.114 | 0.043 | 0.008 | | | |
| Unsafe | -0.224 | 0.061 | 0.000 | | | |
| Very unsafe | -0.375 | 0.112 | 0.001 | | | |
| Job status | | | | | | |
| Student | 0.096 | 0.080 | 0.228 | | | |
| Unemployed, looking for job | -0.555 | 0.114 | 0.000 | | | |
| Unemployed, not looking | -0.407 | 0.146 | 0.005 | | | |
| Permanently sick or disabled | 0.086 | 0.143 | 0.548 | | | |
| Retired | 0.154 | 0.066 | 0.020 | | | |
| Community or military service | 0.259 | 0.284 | 0.361 | | | |
| Housework | 0.033 | 0.066 | 0.610 | | | |
| Other | -0.194 | 0.182 | 0.287 | | | |
| Other factors | | | | | | |
| Years in formal education | 0.006 | 0.006 | 0.298 | | | |
| Intensity of religious belief | 0.053 | 0.009 | 0.000 | | | |
| Belongs to discriminated group | -0.514 | 0.093 | 0.000 | | | |
| Homeowner | 0.097 | 0.045 | 0.030 | | | |
| Marginal effect of deviation from the median in values (baseline = Cluster 1) | | | | | | |
| Cluster 2 | 1.934 | 0.542 | 0.000 | | | |
| Cluster 3 | -1.393 | 1.068 | 0.192 | | | |
| Cluster 4 | 3.662 | 0.823 | 0.000 | | | |
| Model fit statistics: | 1 | | | | | |
| Prob > Chi Square (Wald) | | 0.000 | | | | |
| Pseudo R ² | | 0.059 | | | | |

Table A.7: Ordered logit estimates for happiness: clusters (cont.)

+ Standard errors are adjusted for clustering at the regional level.

RECENTLY PUBLISHED "TEMI" (*)

- N. 643 *The producer service sector in Italy: Long-term growth and its local determinants*, by Valter Di Giacinto and Giacinto Micucci (September 2007).
- N. 644 Aggregazioni bancarie e specializzazione nel credito alle PMI: peculiarità per area geografica, by Enrico Beretta and Silvia Del Prete (November 2007).
- N. 645 *Costs and benefits of creditor concentration: An empirical approach*, by Amanda Carmignani and Massimo Omiccioli (November 2007).
- N. 646 Does the underground economy hold back financial deepening? Evidence from the Italian credit market, by Giorgio Gobbi and Roberta Zizza (November 2007).
- N. 647 *Optimal monetary policy under low trend inflation*, by Guido Ascari and Tiziano Ropele (November 2007).
- N. 648 Indici di bilancio e rendimenti di borsa: un'analisi per le banche italiane, by Angela Romagnoli (November 2007).
- N. 649 *Bank profitability and taxation*, by Ugo Albertazzi and Leonardo Gambacorta (November 2007).
- N. 650 *Modelling bank lending in the euro area: A non-linear approach*, by Leonardo Gambacorta and Carlotta Rossi (November 2007).
- N. 651 Revisiting poverty and welfare dominance, by Gian Maria Tomat (November 2007).
- N. 652 *The general equilibrium effects of fiscal policy: Estimates for the euro area,* by Lorenzo Forni, Libero Monteforte and Luca Sessa (November 2007).
- N. 653 *Securitisation and the bank lending channel*, by Yener Altunbas, Leonardo Gambacorta and David Marqués (November 2007).
- N. 654 The cyclical response of fiscal policies in the euro area. Why do results of empirical research differ so strongly?, by Roberto Golinelli and Sandro Momigliano (January 2008).
- N. 655 What's behind "inflation perceptions"? A survey-based analysis of Italian consumers, by Paolo Del Giovane, Silvia Fabiani and Roberto Sabbatini (January 2008).
- N. 656 *The effects of fiscal policy in Italy: Evidence from a VAR model*, by Raffaela Giordano, Sandro Momigliano, Stefano Neri and Roberto Perotti (January 2008).
- N. 657 *Excess money growth and inflation dynamics*, by Barbara Roffia and Andrea Zaghini (January 2008).
- N. 658 *R&D and market structure in a horizontal differentiation framework*, by Davide Fantino (January 2008).
- N. 659 *Housing market spillovers: Evidence from an estimated DSGE model*, by Matteo Iacoviello and Stefano Neri (January 2008).
- N. 660 *Real exchange rate volatility and disconnect: An empirical investigation*, by Riccardo Cristadoro, Andrea Gerali, Stefano Neri and Massimiliano Pisani (April 2008).
- N. 661 The effect of investment tax credit: Evidence from an atypical programme in Italy, by Raffaello Bronzini, Guido de Blasio, Guido Pellegrini and Alessandro Scognamiglio (April 2008).
- N. 662 Accounting for sampling design in the SHIW, by Ivan Faiella (April 2008).
- N. 663 Delayed privatization, by Bernardo Bortolotti and Paolo Pinotti (April 2008).
- N. 664 *Portfolio selection with mononotone mean-variance preferences*, by Fabio Maccheroni, Massimo Marinacci, Aldo Rustichini and Marco Taboga (April 2008).
- N. 665 Directed matching with endogenous Markov probability: Clients or competitors?, by Emanuela Ciapanna (April 2008).
- N. 666 What are borders made of? An analysis of barriers to European banking integration, by Massimiliano Affinito and Matteo Piazza (April 2008).
- N. 667 Innovation driven sectoral shocks and aggregate city cycles, by Andrea R. Lamorgese (April 2008).
- N. 668 On applying synthetic indices of multidimensional well-being: Health and income inequalities in selected EU countries, by Andrea Brandolini (April 2008).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

L. DEDOLA and F. LIPPI, *The monetary transmission mechanism: Evidence from the industries of 5 OECD countries*, European Economic Review, 2005, Vol. 49, 6, pp. 1543-1569, **TD No. 389 (December 2000)**.

- D. Jr. MARCHETTI and F. NUCCI, *Price stickiness and the contractionary effects of technology shocks*. European Economic Review, Vol. 49, 5, pp. 1137-1164, **TD No. 392 (February 2001)**.
- G. CORSETTI, M. PERICOLI and M. SBRACIA, Some contagion, some interdependence: More pitfalls in tests of financial contagion, Journal of International Money and Finance, Vol. 24, 8, pp. 1177-1199, TD No. 408 (June 2001).
- GUISO L., L. PISTAFERRI and F. SCHIVARDI, *Insurance within the firm*. Journal of Political Economy, Vol. 113, 5, pp. 1054-1087, **TD No. 414** (August 2001)
- R. CRISTADORO, M. FORNI, L. REICHLIN and G. VERONESE, *A core inflation indicator for the euro area,* Journal of Money, Credit, and Banking, Vol. 37, 3, pp. 539-560, **TD No. 435 (December 2001)**.
- F. ALTISSIMO, E. GAIOTTI and A. LOCARNO, *Is money informative? Evidence from a large model used for policy analysis*, Economic & Financial Modelling, Vol. 22, 2, pp. 285-304, **TD No. 445 (July 2002)**.
- G. DE BLASIO and S. DI ADDARIO, *Do workers benefit from industrial agglomeration?* Journal of regional Science, Vol. 45, (4), pp. 797-827, **TD No. 453 (October 2002).**
- G. DE BLASIO and S. DI ADDARIO, Salari, imprenditorialità e mobilità nei distretti industriali italiani, in L.
 F. Signorini, M. Omiccioli (eds.), Economie locali e competizione globale: il localismo industriale italiano di fronte a nuove sfide, Bologna, il Mulino, TD No. 453 (October 2002).
- R. TORRINI, Cross-country differences in self-employment rates: The role of institutions, Labour Economics, Vol. 12, 5, pp. 661-683, TD No. 459 (December 2002).
- A. CUKIERMAN and F. LIPPI, *Endogenous monetary policy with unobserved potential output*, Journal of Economic Dynamics and Control, Vol. 29, 11, pp. 1951-1983, **TD No. 493 (June 2004)**.
- M. OMICCIOLI, Il credito commerciale: problemi e teorie, in L. Cannari, S. Chiri e M. Omiccioli (eds.), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, **TD No. 494 (June 2004)**.
- L. CANNARI, S. CHIRI and M. OMICCIOLI, *Condizioni di pagamento e differenziazione della clientela*, in L. Cannari, S. Chiri e M. Omiccioli (eds.), *Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia*, Bologna, Il Mulino, **TD No. 495 (June 2004)**.
- P. FINALDI RUSSO and L. LEVA, Il debito commerciale in Italia: quanto contano le motivazioni finanziarie?, in L. Cannari, S. Chiri e M. Omiccioli (eds.), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 496 (June 2004).
- A. CARMIGNANI, Funzionamento della giustizia civile e struttura finanziaria delle imprese: il ruolo del credito commerciale, in L. Cannari, S. Chiri e M. Omiccioli (eds.), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 497 (June 2004).
- G. DE BLASIO, Credito commerciale e politica monetaria: una verifica basata sull'investimento in scorte, in L. Cannari, S. Chiri e M. Omiccioli (eds.), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 498 (June 2004).
- G. DE BLASIO, *Does trade credit substitute bank credit? Evidence from firm-level data*. Economic notes, Vol. 34, 1, pp. 85-112, **TD No. 498 (June 2004).**
- A. DI CESARE, *Estimating expectations of shocks using option prices*, The ICFAI Journal of Derivatives Markets, Vol. 2, 1, pp. 42-53, **TD No. 506 (July 2004).**
- M. BENVENUTI and M. GALLO, *Il ricorso al "factoring" da parte delle imprese italiane*, in L. Cannari, S. Chiri e M. Omiccioli (eds.), *Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia*, Bologna, Il Mulino, **TD No. 518 (October 2004)**.
- L. CASOLARO and L. GAMBACORTA, *Redditività bancaria e ciclo economico*, Bancaria, Vol. 61, 3, pp. 19-27, **TD No. 519 (October 2004)**.
- F. PANETTA, F. SCHIVARDI and M. SHUM, *Do mergers improve information? Evidence from the loan market*, CEPR Discussion Paper, 4961, **TD No. 521 (October 2004)**.

2005

- P. DEL GIOVANE and R. SABBATINI, La divergenza tra inflazione rilevata e percepita in Italia, in P. Del Giovane, F. Lippi e R. Sabbatini (eds.), L'euro e l'inflazione: percezioni, fatti e analisi, Bologna, Il Mulino, TD No. 532 (December 2004).
- R. TORRINI, *Quota dei profitti e redditività del capitale in Italia: un tentativo di interpretazione*, Politica economica, Vol. 21, 1, pp. 7-41, **TD No. 551 (June 2005)**.
- M. OMICCIOLI, *Il credito commerciale come "collateral"*, in L. Cannari, S. Chiri, M. Omiccioli (eds.), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, il Mulino, **TD No. 553 (June 2005)**.
- L. CASOLARO, L. GAMBACORTA and L. GUISO, Regulation, formal and informal enforcement and the development of the household loan market. Lessons from Italy, in Bertola G., Grant C. and Disney R. (eds.) The Economics of Consumer Credit: European Experience and Lessons from the US, Boston, MIT Press, **TD No. 560 (September 2005)**.
- S. DI ADDARIO and E. PATACCHINI, *Lavorare in una grande città paga, ma poco*, in Brucchi Luchino (ed.), *Per un'analisi critica del mercato del lavoro*, Bologna , Il Mulino, **TD No. 570 (January 2006)**.
- P. ANGELINI and F. LIPPI, *Did inflation really soar after the euro changeover? Indirect evidence from ATM withdrawals*, CEPR Discussion Paper, 4950, **TD No. 581 (March 2006)**.
- S. FEDERICO, Internazionalizzazione produttiva, distretti industriali e investimenti diretti all'estero, in L. F. Signorini, M. Omiccioli (eds.), Economie locali e competizione globale: il localismo industriale italiano di fronte a nuove sfide, Bologna, il Mulino, **TD No. 592 (October 2002).**
- S. DI ADDARIO, *Job search in thick markets: Evidence from Italy*, Oxford Discussion Paper 235, Department of Economics Series, **TD No. 605 (December 2006)**.

2006

- F. BUSETTI, Tests of seasonal integration and cointegration in multivariate unobserved component models, Journal of Applied Econometrics, Vol. 21, 4, pp. 419-438, **TD No. 476 (June 2003).**
- C. BIANCOTTI, A polarization of inequality? The distribution of national Gini coefficients 1970-1996, Journal of Economic Inequality, Vol. 4, 1, pp. 1-32, **TD No. 487 (March 2004)**.
- L. CANNARI and S. CHIRI, *La bilancia dei pagamenti di parte corrente Nord-Sud (1998-2000)*, in L. Cannari, F. Panetta (a cura di), Il sistema finanziario e il Mezzogiorno: squilibri strutturali e divari finanziari, Bari, Cacucci, **TD No. 490 (March 2004)**.
- M. BOFONDI and G. GOBBI, *Information barriers to entry into credit markets*, Review of Finance, Vol. 10, 1, pp. 39-67, **TD No. 509 (July 2004).**
- FUCHS W. and LIPPI F., *Monetary union with voluntary participation*, Review of Economic Studies, Vol. 73, pp. 437-457 **TD No. 512** (July 2004).
- GAIOTTI E. and A. SECCHI, Is there a cost channel of monetary transmission? An investigation into the pricing behaviour of 2000 firms, Journal of Money, Credit and Banking, Vol. 38, 8, pp. 2013-2038 TD No. 525 (December 2004).
- A. BRANDOLINI, P. CIPOLLONE and E. VIVIANO, *Does the ILO definition capture all unemployment?*, Journal of the European Economic Association, Vol. 4, 1, pp. 153-179, **TD No. 529 (December 2004)**.
- A. BRANDOLINI, L. CANNARI, G. D'ALESSIO and I. FAIELLA, *Household wealth distribution in Italy in the* 1990s, in E. N. Wolff (ed.) International Perspectives on Household Wealth, Cheltenham, Edward Elgar, **TD No. 530 (December 2004)**.
- P. DEL GIOVANE and R. SABBATINI, Perceived and measured inflation after the launch of the Euro: Explaining the gap in Italy, Giornale degli economisti e annali di economia, Vol. 65, 2, pp. 155-192, TD No. 532 (December 2004).
- M. CARUSO, *Monetary policy impulses, local output and the transmission mechanism*, Giornale degli economisti e annali di economia, Vol. 65, 1, pp. 1-30, **TD No. 537 (December 2004).**
- A. NOBILI, Assessing the predictive power of financial spreads in the euro area: does parameters instability matter?, Empirical Economics, Vol. 31, 1, pp. 177-195, **TD No. 544 (February 2005)**.
- L. GUISO and M. PAIELLA, The role of risk aversion in predicting individual behavior, In P. A. Chiappori e C. Gollier (eds.) Competitive Failures in Insurance Markets: Theory and Policy Implications, Monaco, CESifo, **TD No. 546 (February 2005).**

- G. M. TOMAT, Prices product differentiation and quality measurement: A comparison between hedonic and matched model methods, Research in Economics, Vol. 60, 1, pp. 54-68, TD No. 547 (February 2005).
- F. LOTTI, E. SANTARELLI and M. VIVARELLI, *Gibrat's law in a medium-technology industry: Empirical evidence for Italy*, in E. Santarelli (ed.), Entrepreneurship, Growth, and Innovation: the Dynamics of Firms and Industries, New York, Springer, **TD No. 555 (June 2005).**
- F. BUSETTI, S. FABIANI and A. HARVEY, *Convergence of prices and rates of inflation*, Oxford Bulletin of Economics and Statistics, Vol. 68, 1, pp. 863-878, **TD No. 575 (February 2006).**
- M. CARUSO, Stock market fluctuations and money demand in Italy, 1913 2003, Economic Notes, Vol. 35, 1, pp. 1-47, **TD No. 576 (February 2006)**.
- S. IRANZO, F. SCHIVARDI and E. TOSETTI, *Skill dispersion and productivity: An analysis with matched data*, CEPR Discussion Paper, 5539, **TD No. 577 (February 2006).**
- R. BRONZINI and G. DE BLASIO, *Evaluating the impact of investment incentives: The case of Italy's Law* 488/92. Journal of Urban Economics, Vol. 60, 2, pp. 327-349, **TD No. 582 (March 2006).**
- R. BRONZINI and G. DE BLASIO, *Una valutazione degli incentivi pubblici agli investimenti*, Rivista Italiana degli Economisti, Vol. 11, 3, pp. 331-362, **TD No. 582** (March 2006).
- A. DI CESARE, *Do market-based indicators anticipate rating agencies? Evidence for international banks*, Economic Notes, Vol. 35, pp. 121-150, **TD No. 593 (May 2006).**
- L. DEDOLA and S. NERI, What does a technology shock do? A VAR analysis with model-based sign restrictions, Journal of Monetary Economics, Vol. 54, 2, pp. 512-549, TD No. 607 (December 2006).
- R. GOLINELLI and S. MOMIGLIANO, *Real-time determinants of fiscal policies in the euro area*, Journal of Policy Modeling, Vol. 28, 9, pp. 943-964, **TD No. 609 (December 2006).**
- P. ANGELINI, S. GERLACH, G. GRANDE, A. LEVY, F. PANETTA, R. PERLI,S. RAMASWAMY, M. SCATIGNA and P. YESIN, *The recent behaviour of financial market volatility*, BIS Papers, 29, QEF No. 2 (August 2006).

2007

- L. CASOLARO. and G. GOBBI, *Information technology and productivity changes in the banking industry*, Economic Notes, Vol. 36, 1, pp. 43-76, **TD No. 489 (March 2004)**.
- M. PAIELLA, Does wealth affect consumption? Evidence for Italy, Journal of Macroeconomics, Vol. 29, 1, pp. 189-205, TD No. 510 (July 2004).
- F. LIPPI. and S. NERI, *Information variables for monetary policy in a small structural model of the euro area*, Journal of Monetary Economics, Vol. 54, 4, pp. 1256-1270, **TD No. 511 (July 2004)**.
- A. ANZUINI and A. LEVY, *Monetary policy shocks in the new EU members: A VAR approach*, Applied Economics, Vol. 39, 9, pp. 1147-1161, **TD No. 514 (July 2004)**.
- R. BRONZINI, *FDI Inflows, agglomeration and host country firms' size: Evidence from Italy*, Regional Studies, Vol. 41, 7, pp. 963-978, **TD No. 526 (December 2004).**
- L. MONTEFORTE, Aggregation bias in macro models: Does it matter for the euro area?, Economic Modelling, 24, pp. 236-261, **TD No. 534 (December 2004)**.
- A. DALMAZZO and G. DE BLASIO, *Production and consumption externalities of human capital: An empirical study for Italy*, Journal of Population Economics, Vol. 20, 2, pp. 359-382, **TD No. 554 (June 2005).**
- M. BUGAMELLI and R. TEDESCHI, *Le strategie di prezzo delle imprese esportatrici italiane*, Politica Economica, v. 3, pp. 321-350, **TD No. 563 (November 2005)**.
- L. GAMBACORTA and S. IANNOTTI, Are there asymmetries in the response of bank interest rates to monetary shocks?, Applied Economics, v. 39, 19, pp. 2503-2517, **TD No. 566 (November 2005).**
- S. DI ADDARIO and E. PATACCHINI, *Wages and the city. Evidence from Italy*, Development Studies Working Papers 231, Centro Studi Luca d'Agliano, **TD No. 570 (January 2006)**.
- P. ANGELINI and F. LIPPI, Did prices really soar after the euro cash changeover? Evidence from ATM withdrawals, International Journal of Central Banking, Vol. 3, 4, pp. 1-22, TD No. 581 (March 2006).
- A. LOCARNO, Imperfect knowledge, adaptive learning and the bias against activist monetary policies, International Journal of Central Banking, v. 3, 3, pp. 47-85, **TD No. 590 (May 2006)**.

- F. LOTTI and J. MARCUCCI, *Revisiting the empirical evidence on firms' money demand*, Journal of Economics and Business, Vol. 59, 1, pp. 51-73, **TD No. 595 (May 2006)**.
- P. CIPOLLONE and A. ROSOLIA, *Social interactions in high school: Lessons from an earthquake*, American Economic Review, Vol. 97, 3, pp. 948-965, **TD No. 596 (September 2006).**
- A. BRANDOLINI, Measurement of income distribution in supranational entities: The case of the European Union, in S. P. Jenkins e J. Micklewright (eds.), Inequality and Poverty Re-examined, Oxford, Oxford University Press, TD No. 623 (April 2007).
- M. PAIELLA, *The foregone gains of incomplete portfolios*, Review of Financial Studies, Vol. 20, 5, pp. 1623-1646, **TD No. 625 (April 2007).**
- K. BEHRENS, A. R. LAMORGESE, G.I.P. OTTAVIANO and T. TABUCHI, *Changes in transport and non transport costs: local vs. global impacts in a spatial network*, Regional Science and Urban Economics, Vol. 37, 6, pp. 625-648, **TD No. 628** (April 2007).
- G. ASCARI and T. ROPELE, *Optimal monetary policy under low trend inflation*, Journal of Monetary Economics, v. 54, 8, pp. 2568-2583, **TD No. 647** (November 2007).
- R. GIORDANO, S. MOMIGLIANO, S. NERI and R. PEROTTI, *The Effects of Fiscal Policy in Italy: Evidence from a VAR Model*, European Journal of Political Economy, Vol. 23, 3, pp. 707-733, **TD No. 656** (December 2007).

2008

- S. MOMIGLIANO, J. Henry and P. Hernández de Cos, *The impact of government budget on prices: Evidence from macroeconometric models*, Journal of Policy Modelling, v. 30, 1, pp. 123-143 **TD No. 523** (October 2004).
- P. DEL GIOVANE, S. FABIANI and R. SABATINI, What's behind "inflation perceptions"? A survey-based analysis of Italian consumers, in P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 655 (January 2008).

FORTHCOMING

- S. SIVIERO and D. TERLIZZESE, *Macroeconomic forecasting: Debunking a few old wives' tales*, Journal of Business Cycle Measurement and Analysis, **TD No. 395 (February 2001)**.
- P. ANGELINI, *Liquidity and announcement effects in the euro area*, Giornale degli economisti e annali di economia, **TD No. 451 (October 2002).**
- S. MAGRI, Italian households' debt: The participation to the debt market and the size of the loan, Empirical Economics, **TD No. 454 (October 2002)**.
- P. ANGELINI, P. DEL GIOVANE, S. SIVIERO and D. TERLIZZESE, Monetary policy in a monetary union: What role for regional information?, International Journal of Central Banking, TD No. 457 (December 2002).
- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, **TD No. 458 (December 2002).**
- L. GUISO and M. PAIELLA,, *Risk aversion, wealth and background risk*, Journal of the European Economic Association, **TD No. 483 (September 2003).**
- G. FERRERO, *Monetary policy, learning and the speed of convergence*, Journal of Economic Dynamics and Control, **TD No. 499 (June 2004).**
- F. SCHIVARDI e R. TORRINI, *Identifying the effects of firing restrictions through size-contingent Differences in regulation*, Labour Economics, **TD No. 504 (giugno 2004)**.
- C. BIANCOTTI, G. D'ALESSIO and A. NERI, *Measurement errors in the Bank of Italy's survey of household income and wealth*, Review of Income and Wealth, **TD No. 520 (October 2004)**.
- D. Jr. MARCHETTI and F. Nucci, *Pricing behavior and the response of hours to productivity shocks*, Journal of Money Credit and Banking, **TD No. 524 (December 2004).**
- L. GAMBACORTA, *How do banks set interest rates?*, European Economic Review, **TD No. 542 (February 2005).**
- P. ANGELINI and A. Generale, On the evolution of firm size distributions, American Economic Review, TD No. 549 (June 2005).

- R. FELICI and M. PAGNINI, *Distance, bank heterogeneity and entry in local banking markets*, The Journal of Industrial Economics, **TD No. 557 (June 2005).**
- M. BUGAMELLI and R. TEDESCHI, Le strategie di prezzo delle imprese esportatrici italiane, Politica Economica, **TD No. 563 (November 2005).**
- S. DI ADDARIO and E. PATACCHINI, *Wages and the city. Evidence from Italy*, Labour Economics, **TD No. 570** (January 2006).
- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD** No. 578 (February 2006).
- PERICOLI M. and M. TABOGA, Canonical term-structure models with observable factors and the dynamics of bond risk premia, **TD No. 580 (February 2006).**
- E. VIVIANO, Entry regulations and labour market outcomes. Evidence from the Italian retail trade sector, Labour Economics, **TD No. 594 (May 2006)**.
- S. FEDERICO and G. A. MINERVA, *Outward FDI and local employment growth in Italy*, Review of World Economics, Journal of Money, Credit and Banking, **TD No. 613 (February 2007).**
- F. BUSETTI and A. HARVEY, Testing for trend, Econometric Theory TD No. 614 (February 2007).
- V. CESTARI, P. DEL GIOVANE and C. ROSSI-ARNAUD, *Memory for Prices and the Euro Cash Changeover: An Analysis for Cinema Prices in Italy*, In P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, **TD No. 619 (February 2007)**.
- B. ROFFIA and A. ZAGHINI, *Excess money growth and inflation dynamics*, International Finance, **TD No.** 629 (June 2007).
- M. DEL GATTO, GIANMARCO I. P. OTTAVIANO and M. PAGNINI, Openness to trade and industry cost dispersion: Evidence from a panel of Italian firms, Journal of Regional Science, TD No. 635 (June 2007).
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, **TD No. 637 (June 2007).**
- S. MAGRI, *The financing of small innovative firms: The Italian case*, Economics of Innovation and New Technology, **TD No. 640 (September 2007)**.